

POLYNOMIAL BASED RECURSIVE AND NON RECURSIVE FILTER DESIGN

Vinay Kumar

▶ To cite this version:

Vinay Kumar. POLYNOMIAL BASED RECURSIVE AND NON RECURSIVE FILTER DESIGN. Signal and Image processing. Jaypee University of Information Technology, 2009. English. NNT: . tel-00371385

HAL Id: tel-00371385 https://theses.hal.science/tel-00371385

Submitted on 27 Mar 2009

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

THESIS

Polynomial Based Design of Linear Phase Recursive and Non Recursive Filters



submitted by

Vinay Kumar

for the degree of Doctor of Philosophy to the Department of Electronics and Communication Engineering of the Jaypee University of Information Technology, Waknaghat, INDIA

Enrollment Number: 051767

© Copyright by Vinay Kumar, 2008

POLYNOMIAL BASED RECURSIVE AND NON RECURSIVE FILTER DESIGN

ΒY

VINAY KUMAR

THESIS

Submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Department of Electronics and Communication Engineering Jaypee University of Information Technology, 2008

Waknaghat, India

Reviewers of Thesis

- Prof. Manfred Thumm
 Institut f
 ür Hochfrequenztechnik und Elektronik
 Universit
 ät Karlsruhe (TH)
 Geb. 30.10, Kaiserstr 12, D-76128 Karlsruhe
 Germany
- Dr. M V Karthikeyan, Associate Professor Department of Electronics and Computerr Engineering, Indian Institute of Technology Roorkee, India.

Thesis Evaluation Committee

- Dr. M V Karthikeyan, Associate Professor Department of Electronics and Computerr Engineering, Indian Institute of Technology Roorkee, India.
- Prof. T S Lamba
 Department of Electronics and Communication Engineering,
 Jaypee University of Information Technology,
 Waknaghat, Solan-173215,
 India.
- Prof. S V Bhooshan
 HoD, Department of Electronics and Communication Engineering, Jaypee University of Information Technology, Waknaghat, Solan-173215, India.
- 4. Dr. G Singh, Assistant Professor
 Department of Electronics and Communication Engineering,
 Jaypee University of Information Technology,
 Waknaghat, Solan-173215,
 India.
- 5. Prof. H Singh,

HoD, Department of Mathematics, Jaypee University of Information Technology, Waknaghat, Solan-173215, India

Abstract

In this dissertation, several algorithms to design linear phase Finite Impulse Response (FIR) and Infinite Impulse Response (IIR) filters have been discussed.

Contrary to various already existing standard algorithms, the proposed methods approximate magnitude and phase characteristics simultaneously. The basic mechanism used in this study is polynomial based design of digital filters. We have used several already existing polynomials; e.g., Chebyshev polynomials, Legendre polynomials, to develop linear phase digital filters and developed some two dimensional polynomials following orthogonal properties to design digital filters for image processing, their design methodology have also been discussed.

Filters of proposed type can be used for applications where exact linear phase is required. Another application of this type of filters is the design of filters with zero group delay. IIR filters are designed with absolute linear phase and zero group delay.

The algorithms proposed in the present thesis allow user to design filters with his set of constraints, which is required in practical filter design problems. Very narrow band 1D and 2D linear phase FIR filters can easily be designed by the proposed methodology. The IIR filters proposed provide the guarantee to result in a stable filter.

All the algorithms have been discussed stepwise to make sure that any one with basic programming capability can easily design them. We have not used any standard routine of any particular platform, therefore any freely available programming platform (like C, C++, Scilab, Octave, etc.) can be used to design these filters.

Acknowledgment

With deepest gratitude I thank my advisor, Dr. Sunil Vidya Bhooshan, for advising and guiding me throughout the course of this study. He not only guided me through my quest for scientific knowledge, but also encouraged me to explore life itself. I learned much from his enthusiasm toward science. I am forever indebted to him for all that he taught me. Prof. Bhooshan, always there to clear up any of my doubts, helped me through the difficulties at times when I was stuck. Not only is he the best advisor, he is the best friend also.

Many thanks to Prof. G. Singh for his interest in my work, useful remarks and support. He spent his precious time with me discussing about various aspects of my thesis.

I would like to extend a sincere "Thank you" to our vice chancellor Prof. Yajulu Medury for his continuous moral and financial support, without which it was just impossible to take advantage of all the wonderful opportunities. I would like to thank all my colleagues here at JUIT for providing a creative environment in the department. Special thanks to our librarian Sri Ram for his continuous support despite my continuously trouble giving capabilities to him.

Personally, I would like to mention my parents for their continuous support in my education and upbringing. I am thankful to my wife, Shruti, for providing an exceptionally rejuvenating, productive, and enjoyable environment at home. Thanks to Shipra and Monika, my sisters and colleagues, for providing refreshing environment at JUIT and home.

Table of Contents

1	Intr	oduction	1
	1.1	Digital Filters: An Introduction	1
		1.1.1 Ideal Filter Characteristics	3
		1.1.2 Real Filter Characteristics	3
	1.2	Earlier Work in the Field of Digital Filter Design	5
	1.3	Present Work	6
		1.3.1 Transformation	7
	1.4	Outline	12
2	Des	ign of 1 Dimensional Linear Phase FIR Filter with Elemen-	
	tary	Polynomials	15
	2.1	Procedure	15
	2.2	Application	17
		2.2.1 Design 1	17
		2.2.1.1 Calculation of stop band frequency, ω_s , and	
		pass band frequency, $\omega_p \ldots \ldots \ldots$	17
		2.2.1.2 Calculation of b	19
		2.2.2 Design 2	21
	2.3	Conclusion	24
3	Des	ign of 1 Dimensional Linear Phase FIR Filter with Cheby-	
	shev	Polynomials	25
	3.1	Introduction	25
	3.2	Preliminaries	26
	3.3	Procedure	26
	3.4	Application	27
		3.4.1 Design 1	28
		3.4.2 Design 2	30
		3.4.3 Design 3	31
	3.5	Discussion	33
	3.6	Modified Chebyshev Filter	34
	3.7	Application	35
	3.8	Discussion	36
	3.9	Conclusion	36

4	Des	ign of 2 Dimensional Linear Phase FIR Filter with Cheby-
	shev	v Polynomials
	4.1	Introduction
	4.2	Procedure
	4.3	Application
		4.3.1 Design 1
		4.3.2 Design 2
		4.3.3 Design 3
	4.4	Discussion 44
	4.5	Conclusion
F	Daa	ion of 1 Dimensional Linear Dhese EID Filter with Orthog
5	Des	Ign of 1 Dimensional Linear Phase FIK Filter with Orthog-
		I rotynomials
	5.1	
	5.2	Preliminaries
	5.3	Procedure
	5.4	Application and Discussion
		5.4.1 Design 1
		5.4.2 Design 2
		5.4.3 Design 3
		$5.4.4 \text{Design 4} \dots \dots \dots \dots \dots \dots \dots \dots \dots $
	5.5	Results
	5.6	Conclusion
6	Des	ion of 2 Dimensional Linear Phase FIR Filter with Orthoo-
Ū	ona	Polynomials 67
	61	Introduction 67
	6.2	Procedure 68
	63	Application and Discussion 73
	0.0	631 Design 1 74
		6.3.2 Design 2 74
		6.3.2 Design 2
		634 Design 4 79
	61	Resulte 80
	65	Conclusion 86
	0.5	
7	Des	ign of 1 Dimensional Zero or Linear Phase IIR Filter with
	Ort	hogonal Polynomials 95
	7.1	Introduction
	7.2	Preliminaries
	7.3	Procedure
	7.4	Application and Discussion
		7.4.1 Design 1
		7.4.2 Design 2
		7.4.3 Design 3

		7.4.4	Design 4			•••						•				•				108
	7.5	Conclu	usion			•••	•				• •	•		•	• •	•	•	•••	•	109
8	Des	ign of 2	2 Dimens	ional	l Ze	ro o	r L	in	ear	P	ha	se	IIF	R F	ilt	er	w	ith	L	
	Orth	nogona	l Polynon	nials		•••	•			•		•		•	• •	•	•	•••	•	111
	8.1	Introd	uction .									•				•				111
	8.2	Procee	lure									•				•				111
	8.3	Applie	cation and	Disc	cuss	ion						•				•				120
		8.3.1	Design 1			•••						•				•				121
		8.3.2	Design 2			•••	•			•		•				•		•••		124
	8.4	Conclu	usion			•••	•			•	• •	•		•		•	•	•••		134
9	Con	clusion	and Futu	ire N	/ork	••	•	••	•••	•	• •	•		•	• •	•	•		•	143
A	App	oendix	• • • • •	• • •	•••	•••	•	••	••	•	• •	•	••	•	• •	•	•	••	•	145
B	Refe	erences	• • • • •	• • •	•••	•••	•	••	•••	•	• •	•	••	•	• •	•	•	••	•	147
С	Vita	l		• • •		•••	•	••		•		•		•		•	•		•	155

List of Figures

1.1	Typical system to process analog signals via a discrete time	1
10		1
1.2	Low Pass (LP) Inter.	4
1.3	Page d Dage (PD) filter	4
1.4		4
1.5	Flow chart representing steps required to design digital fil-	0
1 (ters following the proposed design technique. \dots	9
1.6	Iransformation equation, $x = x_0 \cos(\omega/2)$, with value of $x_0 = 1$.	10
1./		10
1.8	Object function for filter characteristics of Figure 1.7, show-	11
	ing the slight non linearity in the transition region	11
2.1	Example Polynomial.	16
2.2	Polynomial x^6 .	18
2.3	Magnitude response in <i>dB</i> for Polynomial x^6	20
2.4	Phase response for Polynomial x^6 .	20
2.5	Polynomial $(x - 0)(x - 0.1)(x - 0.2)(x - 0.3)(x - 0.4)(x - 0.5)$.	23
2.6	Magnitude response in dB of $H(z)$ of Equation (2.26)	23
3.1	Magnitude response of 6 th order Chebyshev low pass FIR	
	filter.	29
3.2	Magnitude response in dB of 6 th order Chebyshev low pass FIR filter.	29
3.3	Phase response of 6 th order Chebyshev low pass FIR filter.	30
3.4	Magnitude response of 3 rd order Chebyshev low pass FIR	
	filter.	31
3.5	Magnitude response in dB of 3^{rd} order Chebyshev low pass	
	FIR filter.	32
3.6	Magnitude response of 24 th order Chebyshev low pass FIR	
	filter.	33
3.7	Magnitude response in dB of 24 th order Chebyshev low pass	
	FIR filter.	34
3.8	Magnitude responses of 32 ^{<i>nd</i>} order Chebyshev low pass FIR	
	filter for various values of α	36
/ 1	First faw Chabyshay polynomials in 2 dimension	10
т .1 // Э	Magnitude response in dB of 6^{th} order low pass FIP filter	40 40
т .∠ // 2	Phase response of 6 th order low pass FIR filter	40 42
4.3	1 mase response of 0 of uter row pass fix inter.	43

4.4	Magnitude response in dB of 6 th order Chebyshev high pass FIR filter.	44
4.5	Magnitude response in dB of 20 th order Chebyshev low pass FIR filter.	45
4.6	Magnitude response in dB of 20 th order Chebyshev high pass FIR filter.	45
4.7	Magnitude response in dB of 40^{th} order Chebyshev low pass FIR filter.	46
4.8	Magnitude response in dB of 40 th order Chebyshev high pass FIR filter.	46
4.9	Image-1	47
4.10 4.11	Image-1 passed through Chebyshev low pass filter of 6^{th} order. Image-1 passed through Chebyshev high pass filter of 6^{th}	47
	order.	48
4.12 4 13	Image-2	48
1.10	order.	49
5.1	Plot of first few Legendre polynomials.	53
5.2	Desired characteristics of low pass filter	54
5.3	Object function, for filter characteristics of Figure 5.2, to be	
	approximated using Legendre polynomials.	55
5.4	Approximation of object function using first 10 Legendre	
	polynomials terms, P_0 to P_{18} (the object function here is not	
	showing the non linearities for simplicity).	58
5.5	Magnitude response of the low pass FIR filter corresponding	
	to the object function shown in Figure 5.4.	59
5.6	Magnitude response in dB of the low pass FIR filter corre-	
	sponding to the object function shown in Figure 5.4.	59
5.7	Magnitude response of low pass FIR filter when object func- tion is approximated using 17 Legendre polynomial terms,	
	P_0 to P_{32} .	60
5.8	Magnitude response in dB of low pass FIR filter when object function is approximated using 17 Legendre polynomial	
	terms, P_0 to P_{32} .	61
5.9	Phase response of low pass FIR filter when object function	
	is approximated using 17 Legendre polynomial terms, P_0 to P_{32} .	61
5.10	Magnitude response of high pass FIR filter when object func-	
	tion is approximated using 21 Legendre polynomial terms, P_0 to P_{10} .	62
5 11	Magnitude response in dB of high pass FIR filter when ob-	04
0.11	ject function is approximated using 21 Legendre polynomial	
	terms. P_0 to P_{40}	63
		00

5.12	Magnitude response of band pass FIR filter when object function is approximated using 21 Legendre polynomial terms, P_0 to P_{40} .	64
5.13	Magnitude response in dB of band pass FIR filter when object function is approximated using 21 Legendre polynomial terms, P_0 to P_{40} .	64
5.14	Magnitude response in dB of low pass FIR filter when object function is approximated using 21 Legendre polynomial terms, P_0 to P_{40} .	65
6.1	First few 2D orthogonal polynomials calculated using Equa- tion (6.1).	69
6.2	3D desired filter response.	70
6.3	1D representation of a low pass 2D filter characteristics	70
6.4	3D object function for filter characteristics of Figure 6.2, showing slight non linearity in the transition region	71
6.5	1D object function, to be approximated using Legendre polynomials.	71
6.6	Magnitude response of low pass FIR filter when object func- tion is approximated using 8 orthogonal polynomial terms	75
6.7	Cross section of the magnitude response of low pass FIR fil- ter when object function is approximated using 8 orthogonal polynomial terms.	76
6.8	Magnitude response in dB of low pass FIR filter when object function is approximated using 8 orthogonal polynomial terms	77
6.9	Magnitude response of low pass FIR filter when object func- tion is approximated using 15 orthogonal polynomial terms.	77
6.10	Cross section of the magnitude response of low pass FIR filter when object function is approximated using 15 orthogonal polynomial terms.	78
6.11	Magnitude response in dB of low pass FIR filter when object function is approximated using 15 orthogonal polynomial terms.	79
6.12	Magnitude response of high pass FIR filter when object func- tion is approximated using 15 orthogonal polynomial terms.	80
6.13	Cross section of magnitude response of high pass FIR filter when object function is approximated using 15 orthogonal polynomial terms.	81
6.14	Cross section of magnitude response of band pass FIR filter when object function is approximated using 15 orthogonal polynomial terms.	82
	Г / стана селина с -	

6.15	Full view of magnitude response in dB of band pass FIR filter when object function is approximated using 15 orthogonal
(1(
6.10	Image-1
6.17	orthogonal polynomial terms
6.18	Image-1 passed through low pass filter designed using 15orthogonal polynomial terms.85
6.19	Image-2
6.20	Image-2 passed through high pass filter designed using 15 orthogonal polynomial terms
6.21	Image-2 passed through low pass filter designed using 15orthogonal polynomial terms.88
6.22	Image-3
6.23	Image-3 passed through high pass filter designed using 15 orthogonal polynomial terms
6.24	Image-3 passed through low pass filter designed using 15 orthogonal polynomial terms
6.25	Image-4
6.26	Image-4 passed through high pass filter designed using 15 orthogonal polynomial terms
6.27	Image-4 passed through low pass filter designed using 15orthogonal polynomial terms.94
7.1	Desired filter characteristics for LPF to be used as numerator. 97
7.2	Desired filter characteristics for HPF to be used as denomi- nator
7.3	Object function for LPF of Figure 7.1, showing slight nonlinearity in the transition region.98
7.4	Object function for HPF of Figure 7.2, showing slight non linearity in the transition region.
7.5	Magnitude response of low pass IIR filter designed using 10 orthogonal polynomial terms to approximate object function. 104
7.6	Magnitude response in dB of low pass IIR filter correspond- ing to the object function approximated using 10 orthogonal polynomial terms
7.7	Magnitude response of low pass IIR filter corresponding to the object function approximated using 20 orthogonal polynomial terms
7.8	Magnitude response in dB of low pass IIR filter correspond- ing to the object function approximated using 20 orthogonal polynomial terms

7.9	Phase response of low pass IIR filter corresponding to the object function approximated using 20 orthogonal polynomial terms.
7.10	Magnitude response in dB when numerator and denomi- nator object functions are approximated using 10 and 20
7.11	orthogonal polynomial terms, respectively
	orthogonal polynomial terms, respectively
8.1	3D desired low pass filter response
8.2	3D desired high pass filter response
8.3	3D object function for low pass filter of Figure 8.1, showing
8 /	3D object function for high pass filter of Figure 8.2, showing
0.4	the slight non linearity in the transition region
8.5	Example low pass filter with A=0 and B=1100
8.6	Example low pass filter with A=0 and B=2000
8.7	Example low pass filter with A=2000 and B=2000 118
8.8	Example high pass filter with A=1100 and B=0
8.9	Example high pass filter with A=2000 and B=0 119
8.10	Example high pass filter with A=2000 and B=2000 119
8.11	Desired filter characteristics for LPF
8.12	Desired filter characteristics for HPF
8.13	Magnitude response of low pass IIR filter when object func-
011	tions are approximated using 9 orthogonal polynomial terms. 122
0.14	tions are approximated using 9 orthogonal polynomial terms 123
8 1 5	Magnitude response in dB of low pass IIR filter when object
0.10	functions are approximated using 9 orthogonal polynomial
	terms
8.16	Magnitude response in dB of high pass IIR filter when object
	functions are approximated using 9 orthogonal polynomial
-	terms
8.17	Image-1
8.18	Image-1 passed through low pass filter designed using 9
8 10	Image-1 passed through high pass filter designed using 9
0.19	orthogonal polynomial terms
8.20	Magnitude response of low pass IIR filter when object func-
	tions are approximated using 15 orthogonal polynomial terms. 128
8.21	Magnitude response of high pass IIR filter when object func-
	tions are approximated using 15 orthogonal polynomial terms. 128

8.22	Magnitude response in dB of low pass IIR filter when object functions are approximated using 15 orthogonal polynomial	
	terms	9
8.23	Magnitude response in dB of high pass IIR filter when object	
	functions are approximated using 15 orthogonal polynomial	
	terms	9
8.24	Image-1 passed through high pass filter designed using 15	
	orthogonal polynomial terms	0
8.25	Image-1 passed through low pass filter designed using 15	
	orthogonal polynomial terms	1
8.26	Image-2	2
8.27	Image-2 passed through high pass filter designed using 15	
	orthogonal polynomial terms	3
8.28	Image-3	4
8.29	Image-3 passed through high pass filter designed using 15	
	orthogonal polynomial terms	5
8.30	Image-4	6
8.31	Image-4 passed through high pass filter designed using 15	
	orthogonal polynomial terms	7
8.32	Image-5	8
8.33	Image-5 passed through high pass filter designed using 15	
	orthogonal polynomial terms	9
8.34	Image-6	0
8.35	Image-6 passed through high pass filter designed using 15	
	orthogonal polynomial terms	1

1 Introduction

1.1 Digital Filters: An Introduction

The digital filter is a digital system that can be used to filter discretetime signals. The realization of a digital filter is accomplished by burning a program on a reprogrammable device, like Field Programmable Gate Array (FPGA), or by using a software program. Therefore, if we want to change the characteristics of a digital filter, we only have to change the program which defines the circuit. Figure 1.1 shows a typical discrete time filtering system.



Figure 1.1: Typical system to process analog signals via a discrete time filter.

In Figure 1.1, $x_a(t)$ is the analog input signal, $y_a(t)$ is the analog output signal, x[n] is the digitized $x_a(t)$, y[n] is the digitized $y_a(t)$, $H(\exp(j\omega))$ is the transfer function of the filter, C/D is the continuous to digital signal converter and D/C is the digital to continuous signal converter [1].

Why are we interested in designing digital filters? As stated earlier, digital filters are used for processing digital signals. We interact with digital devices (digital TV, mobile, computers, etc.) and in turn with digital signals everyday, in fact all the time. These devices are used to interact with outer world through some form of channel; for example, space for TV and mobiles, and wire or space for computers (to connect with the internet). Channel introduces noise in the signal due to various well known reasons. These digital filters are used to remove the noise from the digital signal received at the receiver end. Filters are also used to enhance certain features of signal, like edges of an image, quality of audio (treble and bass are enhancement of high and low frequency audio signals,

respectively).

Other than everyday life applications digital filters are used for advanced scientific applications too, like for satellite and medical images. Satellite images are analyzed by applying different operations, filtering being one of the most important. Medical images (like CT scan, X-ray, etc.) and videos (like ultrasound video, etc.) are analyzed better by applying digital filters first, because they remove the noise or enhance certain features and produce better quality images and videos to be analyzed by the medical practitioner. In the present work we will show some satellite, medical and other type of images, and the output produced by applying the proposed filters.

Number of applications are increasing where digital signal processing and digital filtering are used; therefore, digital filters have to be applied for various requirements. Hence, there is a need to have designing procedures which can realize filters with required constraints. In the present thesis, we propose algorithms which can be used to design user defined type of FIR and IIR filters.

Digital filter design had been an active area of research for last several decades. One can design a digital filter having linear or non linear phase. With respect to the last statement consider the Fourier transform pair

$$X(\omega)e^{-j\alpha\omega} \Longleftrightarrow x[n-\alpha] \tag{1.1}$$

Equation (1.1) shows that applying a time delay α (in the discrete time domain) is equivalent to a multiplication of the Fourier transform by a factor of exp ($-j\alpha\omega$); which in turn implies the introduction of a linear phase. When we transmit a signal, a time delay is introduced. Linear phase filters delay all frequencies by the same amount, thereby preserving the waveshape maximally. Therefore, linear phase filters introduce no distortion and are preferred over non linear phase filters. All the proposed filters in the present thesis are linear phase. Primarily, there are two types of digital filters

- 1. Finite Impulse Response (FIR) or Non Recursive filters, and
- 2. Infinite Impulse Response (IIR) or Recursive filters.

FIR or non recursive filters are called by this name because to obtain the

current output sample value of the filter it uses only current and past input samples of the signal, and none of the previous output samples of the filter. Since, there is no feedback from the output there is no instability in these filters and hence they are preferred. IIR or recursive filters use past output samples of the filter also, which can cause instability due to feedback [2].

The order of a FIR filter is the number of previous inputs which have to be stored in order to generate a given output. While the order of IIR filter is the largest number of previous input or output values required to compute the current output.

The nature of the problem coupled with the desired frequency response is – generally – the primary factor to decide which type of filter, FIR or IIR, should be used. Because IIR filters use feedback they tend to be more versatile in accurately approximating the desired magnitude response, but they are not able to produce linear phase. On the other hand, it is easier to design a FIR filter having linear phase. Therefore, if our application requires a linear phase then we go for FIR filters. However, if phase is no issue then we can implement any of the two types of filters provided we make sure that the filter is stable [3].

1.1.1 Ideal Filter Characteristics

A signal is made up of one or more number of frequencies. If a filter passes certain frequencies and stops rest of the frequencies of the signal completely, it is called as an ideal filter. Frequencies which the filter allows to pass through it constitute the *pass band* of the filter and rest form the *stop band*. Figures 1.2(a), 1.3(a) and 1.4(a) show such filters.

1.1.2 Real Filter Characteristics

Real world but ideal filters are different from the ideal filters in the sense that in addition to the pass band and stop band they also have *transition band*. The transition band is the band of frequencies which necessarily has to be present because filters with sharp characteristics are unrealizable. Figures 1.2(b), 1.3(b) and 1.4(b) show these type of filters. In the present thesis these type of digital filters are designed.



Figure 1.2: Low Pass (LP) filter.



Figure 1.3: High Pass (HP) filter.



Figure 1.4: Band Pass (BP) filter.

1.2 Earlier Work in the Field of Digital Filter Design

A lot of work has been done in the field of digital filter design. Fettweis [4] gives an overview of past research in this field. We discuss some of the recent works in the present section.

In the recent past various techniques have been developed to design digital filters. Exchange algorithms [5,6], like Remez exchange algorithm, is relatively difficult to understand and apply. Although linear programming [7] produces linear phase FIR filters but the algorithm is very slow, apart from introducing a very large number of parameters to be optimized. In fact a literature survey shows that all the algorithms which have been proposed suffer from this defect.

To reduce the time requirements one can use semi-infinite programming approach [8], but this technique does not reduces the time requirement drastically. Optimization methods [9, 10] work like the curve fitting technique. They are iterative methods, therefore involve a great deal of computation. To apply optimization techniques one has to give a complete description of the transfer function, which increases the complexity and computation time depending on either the number of poles and zeros, or the coefficients of the transfer function. Optimization technique further suffers from the defect that a global optimum point may not be reached.

In the Chebyshev approximation [11, 12] technique, pass band loss oscillates between A_{min} and A_{max} . If one wants to put limit on stop band also, then he has to use elliptic approximation techniques [2]. A very well known technique was proposed by Parks and McCllean [5,13] and is used widely to get the optimized Chebyshev digital filters.

Weighted Chebyshev approximation [14] uses a point matching technique and generates an error function. The error function is formulated for the desired filter in terms of a linear combination of cosine functions and is then minimized using Remez exchange algorithm [15, 16]. Therefore, this methods suffers from the same defects as we have outlined earlier.

The problems with these methods motivated us to propose a new method to design digital filters.

1.3 Present Work

The present thesis is focused on conceptualizing a technique which will give us a new method of designing digital filters, both recursive and non recursive. The method does not use any optimization technique, which requires recursion, therefore it does not suffer from the defects of computational time requirements. The approach proposed in this thesis is entirely new. Some of the algorithms discussed are based on the design of antenna patterns [17]. Surprisingly this cross disciplinary approach gives very good results.

We propose a polynomial based approach to design digital filters. Although "polynomial based" approaches have been discussed by other authors earlier, yet they are completely different from the present one. One prevailing method for optimized design for FIR filters is Parks-McClellan [18,19] algorithm. The Parks-McClellan algorithm is based on considering filter design problem as a problem in polynomial approximation. In this approach an approximation error function is calculated between desired frequency response and L^{th} order polynomial (in cos ω) approximation of frequency response. Various techniques have been proposed to calculate the optimal solution for this error function [20]. Another method was proposed by Hofstetter, Oppenheim, and Siegel [21] for designing maximal ripple filters. This algorithm is an iterative procedure for producing a polynomial $H^{\star}(e^{j\omega})$ that has extrema of desired values. But the number of iterations depend on the initial guess. Similarly, there are other methods [22,23] too in the litrature, but they also have no relationship with the present work. The method proposed in the present thesis neither requires any optimization technique to get the result nor any guess which has to be minimized iteratively.

The filters proposed in the present thesis are easy to implement and understand, produce excellent magnitude response and absolute linear phase without any approximation. To date there is no optimal algorithm to design absolute zero group delay IIR filters, the present dissertation proposes a technique which results in an absolute zero group delay IIR filter.

The major contribution of the present thesis is that we can design a digital filter, FIR or IIR, having user defined requirements of pass band

and stop band with linear phase. In other words, we can design a low pass, high pass, bandpass, band reject, multiband, etc. type of filters with linear phase. Because of this feature the present design procedure can be applied for a wide range of problems. The technique is extended to design filters for processing images; that is, 2 dimensional digital recursive and non recursive filters.

The outline of the procedure to design a digital filter with desired filter characteristics is as follows:

- 1. First transform the filter characteristics to a function, which we call as object function, by using a special transformation.
- 2. This object function is thereafter realized by using a previously defined set of polynomials.
- 3. The realized object function is then converted back to filter characteristics using inverse of the transform used in step 1.

Therefore, to design a filter we need to understand the transformation, which will be used to frame the object function and later its inverse to realize the filter characteristics itself. In the succeeding section we discuss the transformation.

1.3.1 Transformation

In general, transformations (like bilinear transformation) are used to convert analog filters to digital filters [24] or are used to convert one type of filter to other type of filters [25–27]; for example, low pass filter to band pass filter, low pass to high pass filter. Whereas, the transformation we are going to use in the present work has a completely different function.

The present transformation and its inverse is used to design the digital filters in the present discussion. Figure 1.5 depicts general steps required to design the desired filter characteristics. It must be clear from the Figure 1.5 that first the inverse of the transformation (Equation (1.2)) is used to find the object function, and then after approximation of the object function transformation is used to get the desired filter characteristics. This transformation is used in antenna theory [17] and we apply it to design signal processing filters.

The transformation used in the procedure is

$$x = x_0 \cos(\omega/2) \tag{1.2}$$

where, x_0 is the maximum value of x, it will be clear how x_0 is chosen. Figure 1.6 shows the transformation under discussion. The inference which we can make from Equation (1.2) about the relationship between ω and x is as follows

- when $\omega \in [0, \pi]$ then $x \in [x_0, 0]$, and
- when $\omega \in [-\pi, 0]$ then $x \in [0, x_0]$.

From the properties of Fourier transform we know that the signal spectrum occupies a range of 2π . We assume that $\omega \in [-\pi, \pi]$ in the present work, wherever it is not explicitly mentioned.

Suppose we need to design a filter with characteristics shown in Figure 1.7. First, we apply the inverse of the transformation defined in Equation (1.2) on the filter characteristics – $H(\omega)$ – and it results in an object function – f(x) – as shown in Figure 1.8. Object function thus calculated and the desired filter definition are inverse of each other; or in other words, the maximum values of filter characteristics – pass band – will be mapped to minimum values in object function and vice versa. Similarly, transition region with positive slope becomes a ramp having negative slope and vice versa in object function. This becomes clear if we look at Figures 1.7 and 1.8, which show desired filter characteristics and corresponding object function, respectively. The transformation defined, Equation (1.2), is non linear. Therefore, linearly increasing values of ω are mapped onto non linearly increasing values of x.

Digital filters are defined in terms of delay elements, or z, as they have to be realized using digital storage devices. The transformation of Equation (1.2) gives us filter characteristics directly in terms of z, as shown below



Figure 1.5: Flow chart representing steps required to design digital filters following the proposed design technique.



Figure 1.6: Transformation equation, $x = x_0 \cos(\omega/2)$, with value of $x_0 = 1$.



Figure 1.7: Desired filter characteristics.



Figure 1.8: Object function for filter characteristics of Figure 1.7, showing the slight non linearity in the transition region.

$$x = x_0 \cos(\omega/2)$$

$$\frac{x}{x_0} = \cos(\omega/2)$$

$$= \frac{e^{j\omega/2} + e^{-j\omega/2}}{2}$$

$$= \frac{z^{1/2} + z^{-1/2}}{2} \quad (\text{replacing } e^{j\omega} \text{ with } z) \quad (1.3)$$

Therefore, the object function – defined in terms of x – can directly be converted to delay elements, in terms of z. But in the present form they can not be realized (creating a delay element producing 1/2 delay is not possible). We know that filter characteristics in frequency domain always have even symmetry, therefore, while approximating the object function we use only even power terms of variable which constitutes the polynomial, like $x^0, x^2, x^4, ...$ (this point will be clear in the successive chapters where we design actual filters). Since we shall be using only even power terms, Equation (1.3) will always appear as

$$\left[\frac{z^{1/2} + z^{-1/2}}{2}\right]^{2n}$$

or,

$$\left[\frac{z}{4} + \frac{z^{-1}}{4} + \frac{1}{2}\right]^n$$

Hence, we always have our filter response in terms of z and z^{-1} and can be realized.

This transformation will be used in the successive chapters as the fundamental transformation, if there is any variation in the transform it will be duly explained in the respective chapter.

1.4 Outline

The present thesis is organized in 9 chapters, including the present one. Each of the following chapters, except Chapter 9, propose a new technique for designing digital filters. The organization of the thesis is as follows

Chapter 2 discusses a method where the object function is designed using simple or elementary polynomials. This chapter, primarily, shows the relationship between object function and filter characteristics with the help of simple polynomials.

Chapter 3 presents a design technique which uses Chebyshev polynomials to construct the object function. It also discusses the fundamental limitations of this technique.

Chapter 4 is a filter design technique where 2 dimensional (2D) filter designed by using 2D Chebyshev polynomials.

Chapter 5 gives a design procedure by which user specific filter can be designed by the help of orthogonal polynomials.

Chapter 6 is for designing 2D filters with the help of orthogonal polynomials.

Chapter 7 gives a way to design IIR filters with zero phase, using orthogonal polynomials.

Chapter 8 address the issue of designing 2DIIR filters with zero phase.

Chapter 9 is the concluding chapter of thesis it also addresses future scope of the work.

Let us move to have a look on the design technique to generate *new type* of digital filters.

2 Design of 1 Dimensional Linear Phase FIR Filter with Elementary Polynomials

2.1 Procedure

In this chapter we present an approach to realize FIR filters using elementary polynomials. A stepwise description to design the FIR filter is discussed below

Step 1. Choose a polynomial, f(x), with following properties

- 1. It should have zeros near the origin; that is, $0 \le |x| \le 1$, and
- 2. It should increase sharply for real values of *x*, which are far away from *x* = 1; that is, *x* > 1.

Such a polynomial is shown in Figure 2.1. From figure we can comprehend that as x varies from 0 to some point x_0 , function f(x) traces out a pattern of several side bands and one major band. The side bands are 1/b times down from the major band. This ratio can be selected at will by choosing the value of x_0 .

Step 2. We use the transformation discussed in Chapter 1 for mapping the polynomial variable x to frequency variable ω of filter characteristics. The transformation is repeated below for convenience

$$x = x_0 \cos(\omega/2) \tag{2.1}$$

where, x_0 represents the maximum value of x. At the value when x is x_0 the polynomial has the value b, Figure 2.1.

Step 3. The zeros of the transfer function of FIR filter are calculated next. Let these zeros be represented by $z_i = \exp(j\omega_i)$, where i = 1, 2, ...n. The ω_i s are calculated using the inverse of the transformation of Equation (2.1); that is,
$$\omega_i = 2 \cos^{-1}\left(\frac{x_i}{x_0}\right) \tag{2.2}$$

Note that x_i s are the zeros of the original polynomial f(x). The FIR filter transfer function is given by

$$H(z) = \prod_{i=1}^{n} (z - z_i)$$
(2.3)

or,

$$H(\omega) = \prod_{i=1}^{n} (e^{j\omega} - e^{j\omega_i})$$
(2.4)



Figure 2.1: Example Polynomial.

It will be clear in the next section, where we consider some specific cases of filter design, that the values of x_0 and b are related to the bandwidth of the filter and the ripples in the stop-band, respectively.

2.2 Application

In this section, we consider different types of object functions which, in turn, will give us low pass filters having different magnitude and phase characteristics. Standard frequency transformation routines [2] can be applied on the resulting low pass filter to get the other type of filters; that is, high pass, band pass, etc. The examples below give a perspective of design from object function's side, but once one understood this he can design the filter other way round as well.

2.2.1 Design 1

Let us consider a polynomial where the values of x_i 's are all equal to 0; that is, all the zeros of the object function lie on the origin. In this case the polynomial is

$$f(x) = x^n \tag{2.5}$$

This polynomial satisfies all the conditions specified in *Step 1*. Figure 2.2 shows this polynomial for n = 6.

The value of the function, where we want our stop band to start, can be taken at x = 1. This value is arbitrary, but x = 1 gives good results. Therefore, at the start of the stop band

$$x^n = 1 \tag{2.6}$$

Looking at the transformation $x = x_0 \cos(\omega/2)$ we observe that x = 0 transforms to $\omega = \pi$, x = 1 transforms to $\omega = 2 \cos^{-1}(1/x_0)$ which is the frequency where the stop-band starts.

2.2.1.1 Calculation of stop band frequency, ω_s , and pass band frequency, ω_p

Using above mentioned ideas, we desire that at $x = x_0$ the value of the function $f(x_0)$ should be 'b' times its value than it has at the stop band. Thus,

$$x_0^n = b \tag{2.7}$$



Figure 2.2: Polynomial x^6 .

$$x_0 = b^{(1/n)} \tag{2.8}$$

We use the transform, Equation (2.1), to calculate stop band frequency, ω_s . The stop band starts when the value of *x* is 1, as discussed above, and $x_0 = b^{1/n}$. Therefore, Equation (2.1) converts to

$$1 = b^{(1/n)} \cos(\omega_s/2) \tag{2.9}$$

and,

$$\omega_s = 2\cos^{-1}(1/b^{(1/n)}) \tag{2.10}$$

The pass band frequency, ω_p , of filter characteristic occur when function is $b/\sqrt{2}$ times, or 3 dB down, of its peak value than that at the stop band. Thus,

$$x_p^n = b/\sqrt{2} \tag{2.11}$$

and,

$$x_p = \frac{b^{1/n}}{2^{1/2n}} \tag{2.12}$$

where, x_p is the point on the object function which corresponds to the pass band frequency, ω_p .

Using the transform, Equation (2.1), with Equation (2.12) we can calculate the value of pass band, ω_p , as follows

$$\frac{b^{1/n}}{2^{1/2n}} = b^{1/n} \cos(\omega_p/2) \tag{2.13}$$

therefore,

$$\omega_p = 2 \cos^{-1}(1/2^{1/2n}) \tag{2.14}$$

All the zeros of f(x) lie at x = 0, so there is no need of calculation involved in *Step 3* in this case.

2.2.1.2 Calculation of *b*

As discussed above the maximum value of the object function is b times its value than that at the stop band. Thus, if we want our stop band to be, let us say, p dB down when compared with the maximum value of pass band, the value of b is absolute value of p dB; that is,

$$b = 10^{p/20} \tag{2.15}$$

Suppose in the polynomial discussed above, the value of *n* is 6; that is, our filter is of the order of 6. We want stop band to be 40 *dB* down; that is, $b = 10^{40/20} = 100$ (Equation (2.15)). The value of ω_s is 2.17622 radians and ω_p is 0.6733 radians, calculated using Equations (2.10) and (2.14), respectively. The transfer function for this FIR filter, where $\omega_i = \pi$ and i = 1, 2, ...6, is

$$H(\omega) = \{x_0 \cos(\omega/2)\}^6$$
 (2.16)

The polynomial, magnitude response in dB, and phase response for low pass FIR filter of Equation (2.16) is shown in Figures 2.2, 2.3 and 2.4, respectively. As expected the phase comes out to be zero. The magnitude characteristics is not very good but it gives us an insight into – what our transformation is going to do with the object function after application.



Figure 2.3: Magnitude response in dB for Polynomial x^6 .



Figure 2.4: Phase response for Polynomial x^6 .

2.2.2 Design 2

Let us consider another polynomial with zeros not concentrated at the origin. The polynomial is defined in Equation (2.17) and it follows the characteristics discussed in *Step 1*, as well as it has zeros at locations other than x = 0 except one (Figure 2.5).

$$f(x) = (x - 0)(x - 0.1)(x - 0.2)(x - 0.3)(x - 0.4)(x - 0.5)$$
(2.17)

Order of the filter is 6. Suppose we want our stop band to be 40 dB down; that is, b = 100.

We calculate the value of stop band for the above polynomial by equating, as in previous case, f(x) to 1; that is,

$$(x-0)(x-0.1)(x-0.2)(x-0.3)(x-0.4)(x-0.5) = 1$$
(2.18)

When we solve Equation (2.18) we get six different values of x. Any of the values of x can be used, apart from the fact that every value will give rise to a different value of the frequency where the stop band ends, we choose 1.2646, this point correspond to ω_s and we call it by name x_s .

To calculate the value of x_0 we equate f(x) to b (=100); that is,

$$(x-0)(x-0.1)(x-0.2)(x-0.3)(x-0.4)(x-0.5) = 100$$
(2.19)

The solution of Equation (2.19) gives us six different values of x, here again we consider any one positive value amongst them (see Figure 2.1), we retain $x_0 = 2.4112$.

The value of stop band, ω_s , can be calculated by using the transform discussed in *Step 2*; that is,

$$1.2646 = 2.41121 \cos(\omega_s/2) \tag{2.20}$$

$$\omega_s = 2.0374 \,\mathrm{radians} \tag{2.21}$$

Similarly, the value of pass band, ω_p , can be calculated by equating the

polynomial to $b/\sqrt{2}$ or 100/ $\sqrt{2}$; that is,

$$(x-0)(x-0.1)(x-0.2)(x-0.3)(x-0.4)(x-0.5) = 100/\sqrt{2}$$
(2.22)

Above equation gives us six different values, we consider any one value. In this case every value will correspond to a different value of the frequency where the stop band starts; that is, 3 dB down from the maximum value of the pass band of the filter. The value of the solution is $x_p = 2.29069$. We use the same transformation as discussed in *Step 2* to get the pass band frequency; that is, ω_p

$$2.29069 = 2.41121 \cos(\omega_p/2) \tag{2.23}$$

$$\omega_p = 0.6350 \,\mathrm{radians} \tag{2.24}$$

We follow *Step 2* for the transformation and *Step 3* to calculate the position of zeros for this polynomial. The zeros (in radians) for this polynomial are calculated from Equation (2.2):

$$\omega_{1} = 2\cos^{-1}\frac{0}{x_{0}} = \pi, \quad \omega_{2} = 2\cos^{-1}\frac{0.1}{x_{0}} = 3.0586,$$

$$\omega_{3} = 2\cos^{-1}\frac{0.2}{x_{0}} = 2.9755, \quad \omega_{4} = 2\cos^{-1}\frac{0.3}{x_{0}} = 2.8921,$$

$$\omega_{5} = 2\cos^{-1}\frac{0.4}{x_{0}} = 2.8083, \quad \omega_{6} = 2\cos^{-1}\frac{0.5}{x_{0}} = 2.7238.$$
(2.25)

The transfer function for this FIR filter is (*Step2*)

$$H(\omega) = x_0 \cos(\omega_1/2) * x_0 \cos(\omega_2/2) * x_0 \cos(\omega_3/2) * x_0 \cos(\omega_4/2) * x_0 \cos(\omega_5/2) * x_0 \cos(\omega_6/2)$$
(2.26)

Figures 2.5 and 2.6 show the polynomial and magnitude response in dB for this low pass FIR filter. The magnitude characteristics in dB show that the transition band starts becoming steep when compared with the previous design. It is notworthy that the phase, in this case, will not be linear because zeros are not symmetrical.



Figure 2.5: Polynomial (x - 0)(x - 0.1)(x - 0.2)(x - 0.3)(x - 0.4)(x - 0.5).



Figure 2.6: Magnitude response in dB of H(z) of Equation (2.26).

2.3 Conclusion

In this chapter an insight into the connection of object function with filter, using the transformation function, is presented. With this technique we can design linear phase FIR filter with user defined values of zeros of the filter function and decide how many dB down the stop band should be with respect to the pass band; that is, the value of b.

3 Design of 1 Dimensional Linear Phase FIR Filter with Chebyshev Polynomials

3.1 Introduction

Due to large number of applications [28,29], several approaches have been proposed for designing a Chebyshev FIR filter. To design a linear phase Chebyshev FIR filter one has to convert the filter design problem into an approximation problem [28,29]. Imposing the linear phase restriction in the pass band it leads to a complex approximation problem. An approach discussed by Cuthbert [30] proposes a technique where real and imaginary parts of the filter frequency response are separately approximated to design a non linear phase FIR filter. This approach was further extended using Remez-exchange algorithm [31,32] by Attikiouzel et al [33] for two real approximations. Linear programming was used by Delves et al [34] to approximate real and imaginary parts simultaneously. These papers in general discuss approximation methods, while the approach discussed in the present chapter gives exact design of FIR filter in Chebyshev sense with linear phase.

In this chapter, we are going to discuss a new approach to realize FIR filters using Chebyshev polynomials. Chebyshev polynomials play a vital role in antenna as well as in signal processing theory. The Dolph-Chebyshev distribution of currents feeding the elements of a linear array, comprising an antenna, gives a sharp main lobe and small side lobes all of which have the same power level [17], we use this concept in the present discussion.

This chapter presents a method by which we can design a filter with linear phase and,

(i) A given pass band to stop band ratio,

(ii) A given pass band to stop band transition, and to some extent

(iii) The frequency band of the pass band.

3.2 Preliminaries

A linear equispaced antenna array with n elements, labeled from left to right gives rise to a far field of [17]

$$|E| = |A_0 e^{j0} + A_1 e^{j\psi} + A_2 e^{j2\psi} + \dots + A_{n-2} e^{j(n-2)\psi} + A_{n-1} e^{j(n-1)\psi}|$$
(3.1)

where,

$$\psi = \beta d\cos\phi + \gamma \tag{3.2}$$

|E| is the magnitude of the far field, $\beta = 2\pi/\lambda$, λ is the free space wavelength, d is the spacing between elements, ϕ is the angle from the normal to the linear array, γ is the progressive phase shift from left to right, and A_0, A_1, A_2, \ldots are complex amplitudes which are proportional to the current amplitudes.

If we substitute $z = e^{j\psi}$ and rewrite Equation (3.1), it leads to

$$H(z) = A_0 + A_1 z + A_2 z^2 + \ldots + A_{n-2} z^{n-2} + z^{n-1}$$
(3.3)

this equation represents an FIR filter. Where, H(z) represents the impulse response of the filter with $z = e^{j\omega}$, and A_0 , A_1 , A_2 ,... represent amplitudes at the corresponding frequencies.

The Chebyshev polynomials are given by

$$T_m(x) = \begin{cases} \cos(m \, \cos^{-1} x) & 0 < |x| < 1\\ \cosh(m \, \cosh^{-1} x) & |x| > 1 \end{cases}$$
(3.4)

3.3 Procedure

Let us assume that the order of the filter, which we intend to design, is *m*. The step vise procedure to design the required FIR filter is as follows:

Step 1 : As discussed earlier the pass band to stop band ratio is user dependent. Therefore, first we calculate the absolute value of attenuation,

defined by the user, in the stop band, *b* (refer Chapter 1),

$$b = 10^{(attenuation in dB)/20}$$
(3.5)

Step **2** : We find the stop band, ω_s , and pass band, ω_p , frequencies following the steps discussed in [17]

$$\omega_s = 2\cos^{-1} \left[\frac{1}{\cosh(\overline{1/m}\cosh^{-1}b)} \right]$$
(3.6)

$$\omega_p = 2\cos^{-1}\left[\frac{\cosh\left\{(1/m)\cosh^{-1}(b/\sqrt{2})\right\}}{\cosh(\overline{1/m}\cosh^{-1}b)}\right]$$
(3.7)

Step 3 : The location of zeros on unit circle, ω_m , are calculated by the following equation, which is discussed in [17]

$$\omega_m = 2\cos^{-1}\left\{\frac{\cos(\omega_k)}{\cosh(\overline{1/m}\,\cosh^{-1}b)}\right\}$$
(3.8)

where, $\omega_k = (2k - 1)\pi/2m$, and k = 1...m.

Step **4** : Applying the relation $z_m = e^{jw_m}$ we can write frequency response, H(z), in *z* domain using Equation (3.3) as follows

$$H(z) = (z - z_1)(z - z_2) \dots (z - z_m)$$
(3.9)

where, $z_1, z_2,...$ are location of zeros of the transfer function H(z). Replacing z by $e^{j\omega}$ and z_m 's by $e^{j\omega_m}$'s in Equation (3.9) we get frequency response in frequency domain as follows

$$H(\omega) = (e^{j\omega} - e^{j\omega_1})(e^{j\omega} - e^{j\omega_2})\dots(e^{j\omega} - e^{j\omega_m})$$
(3.10)

To get a vivid picture of the procedure discussed above, we design some actual filters in the following section.

3.4 Application

Let us design some Chebyshev low pass FIR filters with side band 40 dB down from the pass band; that is, $b = 10^{40/20} = 100$.

3.4.1 Design 1

Let us design a filter with order 6; that is, m = 6. From Equations (3.6) and (3.7) we calculate

$$\omega_s = 1.5732,$$
$$\omega_p = 0.5622.$$

The values of the ω_m 's can be calculated by using Equation (3.8)

 $\omega_1 = 1.64,$ $\omega_2 = 2.0958,$ $\omega_3 = 2.7739,$ $\omega_4 = 3.5093,$ $\omega_5 = 4.1874,$ $\omega_6 = 4.6431.$

From Equation (3.10) we can write $H(\omega)$ as

$$H(\omega) = (e^{j\omega} - e^{j1.64})(e^{j\omega} - e^{j2.0958})(e^{j\omega} - e^{j2.7739})(e^{j\omega} - e^{j3.5093})(e^{j\omega} - e^{j4.1874})(e^{j\omega} - e^{j4.6431})$$

If we plot the filter characteristics; that is, $|H(\omega)|$ verses frequency ω , the results will become clear. The *dark continuous lines* in Figures 3.1, 3.2, and 3.3 show the magnitude response, magnitude response in *dB*, and phase response of above mentioned FIR filter, respectively. Figure 3.1 confirms that the location of ω_s and ω_p are at the points where we have calculated them. Figure 3.2 show that side bands are 40 dB down. From Figure 3.3 it is clear that the phase is linear.



Figure 3.1: Magnitude response of 6th order Chebyshev low pass FIR filter.



Figure 3.2: Magnitude response in dB of 6th order Chebyshev low pass FIR filter.



Figure 3.3: Phase response of 6th order Chebyshev low pass FIR filter.

3.4.2 Design 2

Let us calculate the frequency response of 3^{rd} order filter by using the steps mentioned in previous section. The values of ω_s and ω_p for 3^{rd} order filter are

$$\omega_s = 2.4641,$$
$$\omega_p = 0.9136.$$

Zeros, ω_m 's, for this filter are as follows

$$\omega_1 = 2.5578,$$

 $\omega_2 = 3.1416,$
 $\omega_2 = 3.7254.$

While the *dark continuous lines* in Figure 3.4 represents the magnitude response of 3^{rd} order FIR filter, Figure 3.5 shows the magnitude response in *dB*. The side bands are 40 dB down and the bandwidth of the filter is

increased when compared with Figure 3.2. The transition band is wider than the previous design, or in other words filter will pass some of the non required frequencies. Phase remains linear and can be verified easily.



Figure 3.4: Magnitude response of 3rd order Chebyshev low pass FIR filter.

3.4.3 Design 3

Next we calculate the frequency responses of 24^{th} order filter. Suppose we design the filter with side bands 40 dB down, the values of ω_s and ω_p for the 24^{th} order filter are as follows

$$\omega_s = 0.4380,$$

 $\omega_p = 0.1558,$

Values of ω_m 's are as follows

$$\omega_1 = 0.4568, \ \omega_2 = 0.5861;$$

 $\omega_3 = 0.7831, \ \omega_4 = 1.0088;$



Figure 3.5: Magnitude response in dB of 3^{*rd*} order Chebyshev low pass FIR filter.

 $\omega_5 = 1.2478, \ \omega_6 = 1.4935;$ $\omega_7 = 1.7432, \ \omega_8 = 1.9952;$ $\omega_9 = 2.2488, \ \omega_{10} = 2.5033;$ $\omega_{11} = 2.7584, \ \omega_{12} = 3.0138;$ $\omega_{13} = 3.2694, \ \omega_{14} = 3.5248;$ $\omega_{15} = 3.7799, \ \omega_{16} = 4.0344;$ $\omega_{17} = 4.2879, \ \omega_{18} = 4.5400;$ $\omega_{19} = 4.7896, \ \omega_{20} = 5.0354;$ $\omega_{21} = 5.2743, \ \omega_{22} = 5.5001;$ $\omega_{23} = 5.6970, \ \omega_{24} = 5.8264.$ The *dark continuous lines* in Figure 3.6 represents the magnitude response of this order FIR filter and Figure 3.7 shows the magnitude response in *dB*. When Figure 3.6 is compared with Figures 3.1 and 3.4, we found that the present filter has narrow pass band and sharper transition band while side bands remains 40 dB down. Therefore, to design a narrow band filter we have to increase the order of the filter.



Figure 3.6: Magnitude response of 24th order Chebyshev low pass FIR filter.

3.5 Discussion

It is clear from the Figures 3.1, 3.4 and 3.6 and Figures 3.2, 3.5 and 3.7 that the width of the pass band decreases as the order of filter increases, and the transition band becomes steeper, or in other words, it follows brick-wall or ideal characteristics of a filter more closely.

Figure 3.3 shows the phase response of the 6^{th} order FIR filter designed above, which clearly shows its linear nature. The phase responses of 3^{rd} and 24^{th} order filters is also linear and can be verified easily by following the same steps.



Figure 3.7: Magnitude response in dB of 24th order Chebyshev low pass FIR filter.

By using the procedure discussed, we can not control the bandwidth of the resulting filter with predefined order. In the next section we modify the design procedure to overcome this problem.

3.6 Modified Chebyshev Filter

We introduce a *new* parameter in the Chebyshev polynomial to have some control over the bandwidth, when *m* and *b* are already defined. In the original Chebyshev polynomial (Equation (3.4)) we multiply a new parameter ' α ' with parameter 'x' (α controls the bandwidth of the filter). Thus, Equation (3.4) becomes

$$T_{m}(\alpha x) = \begin{cases} \cos(m \cos^{-1} \alpha x) & 0 < |x| < 1\\ \cosh(m \cosh^{-1} \alpha x) & |x| > 1 \end{cases}$$
(3.11)

When we multiply 'x' with ' α ', it results in a change in ω_s only, while ω_p remains the same. The reason is because ' α ' is present in both numerator

as well as in denominator terms (see Equations (3.6) and (3.7)).

Therefore, new stop band frequency, ω_{s-new} , is

$$\omega_{s-new} = 2 \cos^{-1} \left[1/\alpha \left\{ \cosh(\overline{1/m} \cosh^{-1} b) \right\} \right]$$
(3.12)

and,

$$\omega_{p-new} = \omega_p \tag{3.13}$$

We can calculate the location of zeros for modified Chebyshev polynomials by

$$\omega_{m-new} = 2 \cos^{-1} \left[\cos(\omega_k) / \left\{ \alpha (\cosh(\overline{1/m} \cosh^{-1} b)) \right\} \right]$$
(3.14)

where, $\omega_k = (2k - 1)\pi/2m$, and $k = 0 \dots m$. We can write $H(\omega)$ as

$$H(\omega) = \prod_{m-new=0}^{n} (e^{j\omega} - e^{j\omega_{m-new}})$$
(3.15)

3.7 Application

Plotting the magnitude response of Design 1 – with the new parameter α taken into consideration – we get the magnitude response and magnitude response in dB as shown in Figures 3.1 and 3.2, respectively ("*dash followed by dot*" for values $\alpha > 1$ and "*dots*" for values $\alpha < 1$, respectively). It is evident from figures that the bandwidth of our filter is increased in case of $\alpha > 1$ and the stop band is further down.

Figure 3.3 shows that our modified Chebyshev FIR filter has linear phase characteristics, further it confirms that the filter retains its linear phase even after introducing the new parameter α . Similarly Figures 3.4 and 3.5, Figures 3.6 and 3.7 show the magnitude response and magnitude response in *dB* for 3^{*rd*} and 24^{*th*} order FIR filter having side bands 40*dB* down, respectively. For lower order filter α makes very small difference, this can be verified by looking at Figures 3.4 and 3.5.

It is noteworthy that when α has value less than 1, see Figures 3.1 and 3.6, the magnitude of the stop band is approximately equal to the pass band. Thus we *should not* use filters designed using value of $\alpha < 1$.

3.8 Discussion

Figure 3.8 shows the magnitude response in *dB* for various values of α for 32^{nd} order low pass FIR filter. It is clear from the figure that as we increase the value of our new parameter α , the bandwidth of our filter increases and the level of sidebands goes further down. One should note that the value of α less than 1 will give us a poor design, where pass band and stop band are approximately at the same level, and thus should not be used. The value of α is obtained by trial to obtain the desired bandwidth. A filter was designed and synthesized based on this technique to control burst noise, and it produced good results.



Figure 3.8: Magnitude responses of 32^{nd} order Chebyshev low pass FIR filter for various values of α .

3.9 Conclusion

We can say that the FIR filter design technique discussed in the present chapter can easily be implemented for given sideband specifications and bandwidth. The new parameter α further pulls the side band levels down

and increases the bandwidth, while maintaining the phase linear. By this technique we can design band pass, band reject, multiband and other filters using the standard frequency transformation techniques.

We extend this concept to two dimensions, thus making use of such filters for image processing applications as well, in the next chapter.

4 Design of 2 Dimensional Linear Phase FIR Filter with Chebyshev Polynomials

4.1 Introduction

The design of 2 dimensional (2D) linear phase Chebyshev FIR filters has been discussed previously by some authors. An application of linear programming to design such filters was introduced by Hu and Rabiner [35]. But due to its very high computational requirements Fiasconaro [36] developed a better algorithm. Though this algorithm was an improvement but it was still slow. Lu [37] and, Lu and Hinamoto [38] introduced methods based on semi definite programming (SDP) and sequential quadratic programming (SQP). These methods work quite well except for the fact that the design complexity becomes rather high even for filters of moderate order. The approach we discuss in this chapter presents the Chebyshev filter design with a new purview, it results in a linear phase Chebyshev 2DFIR filter. The computation time is less, as the parameters involved to design the filter are few.

4.2 Procedure

1 dimensional (1D) Chebyshev polynomials are given by

$$T_m(x) = \begin{cases} \cos(m \, \cos^{-1} x) & 0 < |x| < 1\\ \cosh(m \, \cosh^{-1} x) & |x| > 1 \end{cases}$$
(4.1)

We desire to design a 2D linear phase FIR filter and for that we require 2D polynomials. To convert 1D Chebyshev polynomial, defined in Equation (4.1), to 2D Chebyshev polynomial we change the variable x with a new variable ρ , which will represent the Chebyshev polynomial in cylin-



Figure 4.1: First few Chebyshev polynomials in 2 dimension.

drical coordinate system. ρ is a mapping function, in other words, the values of ρ are mapped onto variables *x* and *y* (Cartesian coordinates) by the relationship $\rho = \sqrt{x^2 + y^2}$. Thus, Equation (4.1) with new variable ρ is represented as

$$T_m(\rho) = \begin{cases} \cos(m \, \cos^{-1} \rho) & 0 < |\rho| < 1\\ \cosh(m \, \cosh^{-1} \rho) & |\rho| > 1 \end{cases}$$
(4.2)

Some of these polynomials are shown in Figure 4.1.

We call these polynomials as object functions and use them to design our filter. The design is performed by transforming the object function to filter characteristics using the transformation function in the cylindrical coordinate system, in terms of ρ . The transformation is defined as follows

$$\rho = \rho_0 \cos\left(\frac{\omega}{2}\right) \quad -\pi \le \omega \le \pi \tag{4.3}$$

where ρ_0 is the maximum value of ρ .

It is apparent from the discussion of Chapter 1 that this transformation converts the object function, Equation (4.2), to low pass filter; that is, lower

values of the object function will be converted to higher values in filter characteristics and vice versa.

After transformation is applied, object function (Equation (4.2)) transforms to filter characteristics and is given by

$$H_m(\omega) = \begin{cases} \cos\left[m\,\cos^{-1}\left\{\rho_0\,\cos(\omega/2)\right\}\right] & -\pi < |\omega| < \pi\\ \cosh\left[m\,\cosh^{-1}\left\{\rho_0\,\cosh(\omega/2)\right\}\right] & |\omega| > \pi \end{cases}$$
(4.4)

 ω (Equation (4.4)) is mapped onto a 2D frequency domain – defined by u and v – by the relationship $\omega = \sqrt{u^2 + v^2}$. After mapping Equation (4.4) converts to

$$H_{m}(u,v) = \begin{cases} \cos\left[m \, \cos^{-1}\left\{\rho_{0} \cos\left(\frac{\sqrt{u^{2}+v^{2}}}{2}\right)\right\}\right] & -\pi < |u,v| < \pi\\ \cosh\left[m \, \cosh^{-1}\left\{\rho_{0} \cosh\left(\frac{\sqrt{u^{2}+v^{2}}}{2}\right)\right\}\right] & |u,v| > \pi \end{cases}$$
(4.5)

To realize the filter defined in Equation (4.5) we first have to find out the value of ρ_0 . Following the procedure given in [17] we can state that the value of ρ_0 is given by

$$\rho_0 = \cosh\{\cosh^{-1}(b/m)\}$$

where, *m* is the order of the filter, and *b* is the absolute value of the attenuation and is given by

$$b = 10^{(attenuation in \, dB/20)} \tag{4.6}$$

We use this approach of design to model a filter whose stop band attenuation is user defined. Following the procedure outlined in [17] we can find out the values of stop band, ω_s , and pass band, ω_p , frequencies

$$\omega_s = 2\cos^{-1}\left[\frac{1}{\cosh(\overline{1/m}\cosh^{-1}b)}\right]$$
(4.7)

$$\omega_p = 2\cos^{-1}\left[\frac{\cosh\left\{(1/m)\cosh^{-1}(b/\sqrt{2})\right\}}{\cosh(\overline{1/m}\cosh^{-1}b)}\right]$$
(4.8)

Therefore, to design such filters we need to work out following steps *Step 1* : Calculate the absolute value of attenuation in the stop band, *b*,

Equation (4.6).

Step 2 : Calculate the stop band, ω_s , and pass band, ω_p , frequencies, Equations (4.7) and (4.8).

Step 3 : The location of zeros, ω_m , on unit circle can be calculated by the following equation [17]

$$\omega_m = 2\cos^{-1}\left\{\frac{\cos(\omega_k)}{\cosh(\overline{1/m}\cosh^{-1}b)}\right\}$$
(4.9)

where, $\omega_k = (2k - 1)\pi/2m$, and k = 1...m.

Step 4 : Calculate the value of $H_m(u, v)$ using Equation (4.5).

To make the approach and procedure clear we design some such linear phase FIR filters in the next section.

4.3 Application

Suppose we want to design low pass FIR filters with stop band 40 *dB* down; that is, $b = 10^{40/20} = 100$.

4.3.1 Design 1

Let us design a filter of order 6; that is, m = 6. The values of ω_s and ω_p are calculated by using Equations (4.7) and (4.8) and result in 1.5732 radian and 0.5622 radian, respectively. When we realize the filter characteristics of Equation (4.5) with these values of m and b, and plot the magnitude characteristics they result as shown in Figure 4.2, the side bands are 40 dB down. The phase characteristics are shown in Figure 4.3, it is clear from the figure that phase is linear in pass band and stop band.

If we replace $\rho_0 \cos(\sqrt{(u^2 + v^2)}/2)$ in Equation (4.5) by $\rho_0 \sin(\sqrt{(u^2 + v^2)}/2)$ we get high pass filter, since *sinusoidal* function transforms low values in object function to low values of filter magnitude characteristics and vice versa. This high pass filter is shown in Figure 4.4.

4.3.2 Design 2

We extend our design to a filter whose order is 20; that is, m = 20. Again we follow the steps of previous section and calculate the filter characteristics.



Figure 4.2: Magnitude response in dB of 6th order low pass FIR filter.



Figure 4.3: Phase response of 6th order low pass FIR filter.



Figure 4.4: Magnitude response in dB of 6^{th} order Chebyshev high pass FIR filter.

The low pass and high pass filter magnitude characteristics of this order are shown in Figures 4.5 and 4.6, respectively. The resulted filter has side bands which are 40 dB down and filter pass band becomes narrower when compared with the previous design. Transition bands are also sharper than the previous design.

4.3.3 Design 3

We design a filter of order 40, following the steps outlined in the the previous section. Figures 4.7 and 4.8 show the magnitude response of low and high pass filters of 40th order, respectively. Note the very narrow pass band of resulting filters.

4.4 Discussion

If we look at Figures 4.2, 4.5 and 4.7, and Figures 4.4, 4.6 and 4.8 we can easily state that the pass band of the filters successively become narrower



Figure 4.5: Magnitude response in dB of 20th order Chebyshev low pass FIR filter.



Figure 4.6: Magnitude response in dB of 20th order Chebyshev high pass FIR filter.



Figure 4.7: Magnitude response in dB of 40th order Chebyshev low pass FIR filter.



Figure 4.8: Magnitude response in dB of 40th order Chebyshev high pass FIR filter.

and narrower as we increase the order of the Chebyshev polynomials, which are used to design the filter.

To show the performance of the filters designed by using the proposed methodology in the present chapter, we pass an image through some of these filters. Figure 4.9 shows an image, which we pass through our 6th order filter, Figures 4.10 and 4.11 are low pass and high pass filtered results, respectively. In an image large blurred details are low frequencies, and small, sharp details represent high frequencies. In image of Figure 4.10 only large overview of the image is available for view, while Figure 4.11 shows the sharp details of the original image.



Figure 4.9: Image-1.



Figure 4.10: Image-1 passed through Chebyshev low pass filter of 6th order.

Application of high pass filter becomes more clear if we apply it to an image having a lot of high frequency components. A satellite image is such an image, it consists of a large number of sharp details. Figures



Figure 4.11: Image-1 passed through Chebyshev high pass filter of 6^{th} order.



Figure 4.12: Image-2.



Figure 4.13: Image-2 passed through Chebyshev high pass filter of 6^{th} order.

4.12 and 4.13 show such an image and its high pass filtered outcome when it is passed through a 6th order high pass FIR filter, respectively. The filtered outcome shows, in general, the edges present in the original image. Therefore, we can use this filter to enhance the edges of a satellite image. This sharper image can be used for further analysis, like finding a street or building on a city map. Further, an edge detection algorithm together with the filter may generate excellent results.

4.5 Conclusion

An alternate approach to design 2D linear phase FIR filters, using Chebyshev polynomials, with the help of the transformation has been examined. By this approach filters can directly be designed from defined 2D Chebyshev polynomials. The filters designed are equiripple and as we increase the order of the filter the sideband goes further down, thus improving the filter characteristic by the rejection of parts which are not required. One can design very narrow band filter using the present technique, thus passing only those frequencies which are required (useful for satellite, medical, astronomical, etc. type of images). Any other type of filter can be realized by shifting the pass band of filter using any of the standard frequency transformation techniques.

5 Design of 1 Dimensional Linear Phase FIR Filter with Orthogonal Polynomials

5.1 Introduction

To design linear phase FIR filters, various techniques have been discussed by several authors. Some of them, like Remez exchange algorithm [18] [6], are generally used for efficient implementation of linear phase FIR filters, but as the number of conditions grow the computational time increases exponentially.

Linear programming (LP), semi-infinite programming (SIP) and iterative re-weighted least square (IRLS) algorithms are generally used to design filters with constraints. Liang *et al.* [7] used LP technique to design some filters with constraints, like optimal Nyquist filters, partial response filters, etc. but this method consumes a great deal of CPU time. Potchinkov [8] applied SIP technique for constrained filters but the time requirements were still high. IRLS algorithms by Burrus et al [39] and many others have good computational efficiency. But they are bad when convergence is considered, because it is not guaranteed that they will converge.

Some techniques to design FIR filter are discussed by various authors, where either the filter has good magnitude response but poor phase characteristics or if they possess exact linear phase then magnitude characteristics are poor or they can not be controlled completely. Chan and Tsui [40] discusses an approach where the group delay is not constant and consequently the phase is not linear. Hanna [41] discusses a method where as bandwidth of the filter increases, the level of sidebands increases and also when the cutoff frequency goes far from the origin pass band characteristic becomes less flat. These requirements motivate us to discuss a new technique to design an arbitrary magnitude response and exact linear phase FIR filter using a very simple technique.
5.2 Preliminaries

A sequence of orthogonal polynomials with weight W(x) satisfies the relation

$$\langle f_i(x), f_j(x) \rangle = \int_{x_2}^{x_1} f_i(x) f_j(x) W(x) dx = 0 \qquad i \neq j$$
 (5.1)

where, $f_i(x)$ and $f_j(x)$ are any two members of the orthogonal set, and $\langle \bullet, \bullet \rangle$ represents the inner product.

In the present discussion, we consider only Legendre polynomials as an example of orthogonal polynomials. Since they are the simplest orthogonal polynomials among all – generally used – yet the procedure is general in nature and one can use any of the other orthogonal polynomials and consider their interval of orthogonality.

For Legendre polynomials the interval of orthogonality is [-1,1] and weight function is 1. We can generate the Legendre polynomials by using Rodrigues' Formula [42] given by

$$P_n(x) = \frac{1}{2^n n!} \frac{d^n}{dx^n} (x^2 - 1)^n$$
(5.2)

where, *n*=1, 2, 3, The first few polynomials are defined below:

$$P_0(x) = 1 (5.3)$$

$$P_1(x) = x \tag{5.4}$$

$$P_2(x) = \frac{3x^2 - 1}{2} \tag{5.5}$$

$$P_4(x) = \frac{5x^3 - 3x}{2}$$
(5.6)
:

These all are orthogonal to each other over [-1, 1]; that is

$$\int_{-1}^{1} P_m(x)P_n(x)dx = 0 \qquad \text{whenever } m \neq n \qquad (5.7)$$

The plot of the first few Legendre polynomials is shown in Figure 5.1.



Figure 5.1: Plot of first few Legendre polynomials.

5.3 Procedure

Suppose user needs to design a low pass filter (LPF), whose magnitude response, $H(\omega)$, is shown in Figure 5.2¹ where,

 A_{max} represents the amplitude of the pass band of the filter,

 A_{min} represents the amplitude of the stop band of the filter,

 ω_{OH} is the end of pass band and the start of the transition band, and ω_{OL} is the end of transition band and start of the stop band.

 $H(\omega)$ (Figure 5.2) is first converted to an object function (independent variable *x*), shown in Figure 5.3. The mapping from Figure 5.2 to 5.3 is done by previously defined transformation (repeated below)

$$x = x_0 \cos(\omega/2) \qquad -\pi < \omega < \pi \tag{5.8}$$

The object function is then approximated using a linear combination of several Legendre polynomials². Note that we approximate the object

¹Though here an LPF is considered, any other ideal magnitude characteristic may be used. The procedure is general in nature.

²The proof that any object function can be approximated by a linear combination of

function with even Legendre polynomial terms only; that is, P_0, P_2, P_4, \ldots , since the characteristics of Figure 5.3 are symmetric.

Let us call the object function of Figure 5.3 as f(x), and it is given by

$$f(x) = \sum_{n=0}^{\infty} a_{2n} P_{2n}(x)$$
 (5.9)

where, $a_0, a_2, a_4, ...$ are coefficients which have to be multiplied with the corresponding Legendre polynomials to get the required characteristics, shown in Figure 5.3.



Figure 5.2: Desired characteristics of low pass filter.

Let us discuss the procedure to design the filter step by step.

Step 1 : To calculate the coefficients $(a_0, a_1, a_2, ...)$ we obtain a set of equations by multiplying Equation (5.9) by $P_0(x)$, $P_2(x)$,... one by one and integrating over the interval [-1, 1]³this leads to following array of equations

even powered Legendre polynomial terms is given in Appendix A. ³See Appendix A



Figure 5.3: Object function, for filter characteristics of Figure 5.2, to be approximated using Legendre polynomials.

$$2\int_{0}^{1} f(x)P_{0}(x)dx = 2\int_{0}^{1} \left(\sum_{n=0}^{\infty} a_{2n}P_{2n}(x)\right)P_{0}(x)dx$$
(5.10)

$$2\int_{0}^{1} f(x)P_{2}(x)dx = 2\int_{0}^{1} \left(\sum_{n=0}^{\infty} a_{2n}P_{2n}(x)\right)P_{2}(x)dx$$
(5.11)

$$2\int_{0}^{1} f(x)P_{4}(x)dx = 2\int_{0}^{1} \left(\sum_{n=0}^{\infty} a_{2n}P_{2n}(x)\right)P_{4}(x)dx$$
(5.12)

The number of equations which we have to solve depends on how many coefficients we desire, or in other words how many Legendre polynomials we are going to use for the approximation of our object function, this point will be clear soon.

:

Using the orthogonality property Equations (5.10), (5.11), ... reduce to

$$\int_{0}^{1} f(x)P_{0}(x)dx = \int_{0}^{1} a_{0}(P_{0}(x)P_{0}(x))^{2}dx$$
(5.13)

$$\int_{0}^{1} f(x)P_{2}(x)dx = \int_{0}^{1} a_{2}(P_{2}(x)P_{2}(x))^{2}dx$$
(5.14)
:

From the above equations, it is clear that the coefficients $a_0, a_2, ...$ are calculated directly by using

$$a_{i} = \frac{\int_{0}^{1} f(x)P_{i}(x)dx}{\int_{0}^{1} P_{i}(x)P_{i}(x)dx}$$
(5.15)

where, *i* = 0, 2, 4,

Step 2 : The coefficients are replaced in Equation (5.9), and the resulting approximate polynomial $f_a(x)$ is calculated. Note that $f_a(x)$ has a limited number of terms, up to N, when compared with f(x) Equation (5.9); that is,

$$f_a(x) = \sum_{n=0}^{N} a_{2n} P_{2n}(x)$$
(5.16)

Step 3 : To transform from polynomial (or 'x') domain to frequency (or ' ω ') domain we use the transformation, Equation (5.8). We choose x_0 to be 1, albeit one can choose any value.

Step 4 : The zeros of $f_a(x)$ are calculated using any standard routine and the zeros of $f_a(\omega)$ are obtained by using

$$x_i \xrightarrow{\mathcal{T}} \omega_i$$

here x_i s are the zeros of $f_a(x)$ and \mathcal{T} is the transformation. The zeros of H(z) are $z_i = exp(j\omega_i)$, where $i = 1, 2, 3 \dots$

Step 5 : Using these zeros we calculate the transfer function in *z* domain; that is,

$$H(z) = (z - z_1)(z - z_2)(z - z_3) \dots$$
(5.17)

and replace $z = e^{j\omega}$ to find $H(\omega)$.

5.4 Application and Discussion

Let us design a filter with following values (refer Figure 5.2)

$$A_{max} = 1000,$$

 $A_{min} = 0,$
 $\omega_{OL} = 2.3186,$
 $\omega_{OH} = 2.0007,$
 $\omega_{3dB} = 2.0944.$

Then, the "object function" characteristics are calculated to be

$$f(x) = \begin{cases} 0 & \text{if } 0 \le x < 0.4 \\ 7142.86(x - 0.4) & \text{if } 0.4 \le x < 0.54 \\ 1000 & \text{if } 0.54 \le x < 1.0 \end{cases}$$
(5.18)

Let us consider some specific cases for above mentioned values.

5.4.1 Design 1

As we keep increasing the number of terms to approximate the object function we get better and better results. It is up to the designer, where he would like to stop. For example, when ten even Legendre polynomial terms are used to approximate the object function then

$$f_a(x) = a_0 P_0 + a_2 P_2 + a_4 P_4 + a_6 P_6 + \dots + a_{18} P_{18}$$
(5.19)

where, $f_a(x)$ is the approximation to f(x).

Using Equation (5.15) we calculate the coefficients a_0 to a_{18} , and

$$f_a(x) = 530 + 909.685P_2 - 586.385P_4 - 4.6429P_6 + 401.108P_8 + \dots$$
(5.20)

The approximated, together with ideal, object function is shown in Figure 5.4. It is clear from the figure that approximated object function follows

the user defined characteristics, but not very closely (in Design 2 we try to improve this).

Using the transformation described in Step 3, we transform $f_a(x)$ to frequency domain. Step 4 will give the position of the eighteen zeros in the frequency domain.

Following Step 5 we can write H(z) as

$$H(z) = \prod_{i=1}^{18} (z - z_i)$$
(5.21)

where, z_i 's are $e^{j\omega_i}$; i = 1, 2, ... 18. This H(z) gives the magnitude response shown in Figure 5.5 and is following the required filter characteristics, also shown in the figure. Magnitude response in dB is shown in Figure 5.6. If we look at it we found that the pass band is very linear and stop bands are approximately 25dB down from the pass band.



Figure 5.4: Approximation of object function using first 10 Legendre polynomials terms, P_0 to P_{18} (the object function here is not showing the non linearities for simplicity).



Figure 5.5: Magnitude response of the low pass FIR filter corresponding to the object function shown in Figure 5.4.



Figure 5.6: Magnitude response in dB of the low pass FIR filter corresponding to the object function shown in Figure 5.4.

5.4.2 Design 2

Let us approximate our object function of Equation (5.18) using 17 even Legendre polynomial terms. We get the magnitude response and the magnitude response in dB as shown in Figures 5.7 and 5.8, respectively. The object function and thus magnitude response are following the required characteristics more closely than in Design 1 (Figure 5.5), which shows that as we increase the number of terms to approximate our object function we get better approximation of magnitude response. The dB magnitude response (Figure 5.8) shows that the first side band is approximately 35dB down; that is, there is an improvement of 10dB over previous design. Figure 5.9 shows the phase response for the present design, the phase is exactly linear.



Figure 5.7: Magnitude response of low pass FIR filter when object function is approximated using 17 Legendre polynomial terms, P_0 to P_{32} .

5.4.3 Design 3

Following the above procedure we can easily design a high pass filter also. Let us consider the following values for the filter to be designed

$$A_{max} = 1000,$$



Figure 5.8: Magnitude response in dB of low pass FIR filter when object function is approximated using 17 Legendre polynomial terms, P_0 to P_{32} .



Figure 5.9: Phase response of low pass FIR filter when object function is approximated using 17 Legendre polynomial terms, P_0 to P_{32} .

$$A_{min} = 0,$$

 $\omega_{OH} = 2.3186,$
 $\omega_{OL} = 2.0007,$
 $\omega_{3dB} = 2.0944.$

Figures 5.10 and 5.11 show the magnitude response, and magnitude response in dB, respectively, of the high pass filter. The high pass filter is derived from the object function which is approximated using 21 Legendre polynomials.



Figure 5.10: Magnitude response of high pass FIR filter when object function is approximated using 21 Legendre polynomial terms, P_0 to P_{40} .

It is clear from the Figures 5.10 and 5.11 that the pass band of the resulting high pass filter is very flat, phase of this filter is absolutely linear, and the realized filter closely follows the restrictions; that is, cutoff frequency and pass band frequency.



Figure 5.11: Magnitude response in dB of high pass FIR filter when object function is approximated using 21 Legendre polynomial terms, P_0 to P_{40} .

5.4.4 Design 4

Next we design a band pass filter with the transition band, stop band, and pass band characteristics as follows

$$A_{max} = 1000,$$

 $A_{min} = 0,$
 $\omega_{OL1} = 1.5908,$
 $\omega_{OH1} = 2.0944,$
 $\omega_{OL2} = 2.7389,$
 $\omega_{OH2} = 2.5322.$

Note that the filter has unequal transition bands.

The magnitude response, and magnitude response in dB are calculated using 21 Legendre polynomial terms to approximate the corresponding



Figure 5.12: Magnitude response of band pass FIR filter when object function is approximated using 21 Legendre polynomial terms, P_0 to P_{40} .



Figure 5.13: Magnitude response in dB of band pass FIR filter when object function is approximated using 21 Legendre polynomial terms, P_0 to P_{40} .

object function and are shown in Figures 5.12 and 5.13, respectively. Unequal side bands are distinguishable in Figure 5.12. The pass band is not as flat as we need but if we increase the number of terms, used to approximate object function, it will become better. The phase response remains linear and can be verified easily.

5.5 Results

We can conclude that as we increase the number of terms in our polynomial, the approximation of the object function becomes more and more like the desired object function (see Figures 5.5 and 5.7).





Figures 5.6, 5.8, and 5.14 show a steady improvement in all round performance in the magnitude responses. It is evident from the figures that the sharpness of the transition band, side band level and ripple in the pass band are all improved. We can also easily deduce from the figures that the filter characteristics are in accordance with the defined values of ω_{OL} , ω_{OH} and ω_{3dB} .

Figures 5.11 and 5.13 make it clear that the method proposed can easily be implemented to design FIR filters of another kind; that is, high pass, bandpass, as well as band reject, multiband, and any other type, all we have to do is write down the corresponding object function characteristics and follow the procedure discussed.

Important Note: If we approximate the "ideal brick wall characteristics"; that is, with an abrupt change at a particular point in place of a gradual change, the results show serious limitations because of Gibbs type of phenomenon. In the case of an abrupt change the approximation gives us a flat pass band but it turns out that the stop band does not reduce below a certain value irrespective of how ever many terms we use.

5.6 Conclusion

An alternative simple approach has been presented to design the linear phase FIR filters, whose characteristics are modeled using orthogonal polynomials. The orthogonal polynomials give us good frequency domain characteristics in terms of sharp cutoff, low stop band, low ripple in the pass band and linear phase. These may be improved to any desired level by increasing the number of terms used to approximate the object function. Further, there is no restriction on the type of filter to be designed.

6 Design of 2 Dimensional Linear Phase FIR Filter with Orthogonal Polynomials

6.1 Introduction

Due to rapid development in the field of image processing, sonar signal processing and other related fields, the design of 2 dimensional (2D) digital filters has become very important. 2D finite impulse response (FIR) filters are preferred over infinite impulse response (IIR) filters because of their inherent stability. These 2DFIR filters may have linear phase¹.

Various techniques have been discussed in the literature to design 2DFIR digital filters. If we take into consideration least square (LS) error criterion, we observe that an overshoot of the frequency response at the pass band and the stop band edges occur due to Gibb's phenomenon [43]. Whereas, a minmax design approach results in an equiripple solution [44,45].

Efficient 2DFIR filters can also be designed by using transformations. J. H. McClellen [46,47] has proposed a powerful technique, where 1DFIR filters are used with a transformation to design 2DFIR filters. This technique is very efficient and fast. This technique was implemented by [48]. But in this technique the transformation steps are very complicated and difficult to implement, and can be used to design 2D square filters only.

What we are presenting here is a new method to design 2DFIR filter with linear phase through the transformation and orthogonal polynomials. The whole procedure is simple and produce excellent results. Any type of filter can be designed by this approach.

¹It is intuitively clear that non linear phase will introduce a distortion.

6.2 Procedure

Because we are going to design an orthogonal polynomial based 2D FIR filter, therefore first we have to calculate 2D orthogonal polynomials.

We know that a sequence of orthogonal polynomials, $P_n(\rho)$ in cylindrical coordinates, must satisfy the relation

$$\left\langle P_i(\rho), P_j(\rho) \right\rangle = \int_0^{2\pi} \int_0^1 P_i(\rho) P_j(\rho) \rho d\rho d\phi = 0 \qquad i \neq j \qquad (6.1)$$

where,

 $P_i(\rho)$ and $P_j(\rho)$ are any two members of the orthogonal set,

 $\langle \bullet, \bullet \rangle$ represents the inner product, and

 $\rho d\rho d\phi$ is the element of area.

The polynomials can be obtained by applying the well known *Graham*-*Schmidt* procedure [49] over the interval [-1, 1] for ρ . Equation (6.2) represents some of the even polynomials.

$$P_{0}(\rho) = 1,$$

$$P_{2}(\rho) = 1 + 2\rho^{2},$$

$$P_{4}(\rho) = 1 - 6\rho^{2} + 6\rho^{4},$$

$$P_{6}(\rho) = 1 - \frac{192}{31}\rho^{2} + \frac{190}{31}\rho^{4} + \frac{20}{31}\rho^{6}$$

$$\vdots \qquad (6.2)$$

Figure 6.1 shows some of these polynomials.

The filters which we propose to design are circularly symmetric 2DFIR filters, half of the symmetric magnitude response of such a filter is shown in Figure 6.2, other half being mirror image. Since the design is circularly symmetric, we use the cylindrical co-ordinate system – (ρ , ϕ) – to represent the object function in 2D. ρ is a mapping function and is related to Cartesian coordinates (x, y) by

$$\rho^2 = x^2 + y^2 \quad -1 \le \rho \le 1 \tag{6.3}$$

Since the filter is circularly symmetric, therefore, if we take the cross section of the 2D FIR filter perpendicular to u - v plane passing through



Figure 6.1: First few 2D orthogonal polynomials calculated using Equation (6.1).

the origin, it will look like a 1D filter. Figure 6.3 shows the 1D equivalent magnitude response for the filter shown in Figure 6.2. In Figure 6.3

 A_{max} represents the amplitude of the pass band of the filter,

 A_{min} represents the amplitude of the stop band of the filter,

 $\omega_{\rm OH}$ is the end of pass band and start of the transition band, and

 ω_{OL} is the end of transition band and start of the stop band.

The relationship between the 1D frequency axis, ω -axis, of Figure 6.3 and 2D frequency axis, *u*, *v*-axis, of Figure 6.2 is given by

$$\omega^2 = u^2 + v^2 \quad -\pi \le \rho \le \pi \tag{6.4}$$

The transformation used for conversion from object function – ρ domain – to frequency characteristics – ω domain – is carried out by

$$\rho = \rho_0 \cos\left(\frac{\omega}{2}\right) \quad -\pi \le \omega \le \pi \tag{6.5}$$

Because the filter is circularly symmetric, therefore, the object function will also be circularly symmetric around the *z*-axis. Figure 6.4 shows this object function in 3D and its corresponding 1D representation in Figure 6.5.



Figure 6.2: 3D desired filter response.



Figure 6.3: 1D representation of a low pass 2D filter characteristics.



Figure 6.4: 3D object function for filter characteristics of Figure 6.2, showing slight non linearity in the transition region.



Figure 6.5: 1D object function, to be approximated using Legendre polynomials.

To transform the 2D filter function, in terms of u - v, to 2D object function, in terms of x - y, we use

$$u = 2\cos^{-1}(x/\rho_0) \quad \pi \le |u|, \rho_0 < |x|$$

$$v = 2\cos^{-1}(y/\rho_0) \quad \pi \le |v|, \rho_0 < |y|$$
(6.6)

$$f(x,y) \xrightarrow{\mathcal{T}} H(u,v)$$

where, ρ_0 is the maximum value of ρ and \mathcal{T} is the transformation of Equation (6.6), it is also the maximum value of *x* and *y*, Equation (6.3).

We approximate the object function using a linear combination of several orthogonal polynomials ², calculated above. Note that approximation of the object function will be done using even polynomials only; that is, $P_0(\rho), P_2(\rho), \ldots$, because the object function and frequency response both are symmetric in nature, and polynomials with even power will give us the required symmetrical characteristics.

Let us denote the object function of Figure 6.4 by $f(\rho)$ (recall Equation (6.3)), which can be written as

$$f(\rho) = \sum_{n=0}^{\infty} a_{2n} P_{2n}(\rho)$$
(6.7)

where, $a_0, a_2, ...$ are coefficients to be multiplied with orthogonal polynomials to get the required characteristics, shown in Figure 6.4. To calculate these coefficients, $a_0, a_2, ...$, we use ³

$$a_i = \frac{\int_0^1 f(\rho) P_i(\rho) \rho d\rho}{\int_0^1 P_i(\rho) P_i(\rho) \rho d\rho}$$
(6.8)

where, i = 0, 2, 4, ...

We calculate the approximate object function $f_a(\rho)$ by

$$f_a(\rho) = \sum_{i=0}^N a_i P_i(\rho) \tag{6.9}$$

²Any function can be represented as a linear combination of several orthogonal polynomials, refer Appendix A.

³Refer Appendix A.

Note that $f_a(\rho)$ tends to $f(\rho)$ when we use infinite number of terms to approximate the object function. Therefore, as we increase number of terms in $f_a(\rho)$ we get better approximation of $f(\rho)$.

This approximate object function is then transformed from (x, y) or ρ domain to frequency domain, or (u, v) or ω domain, by using Equation (6.5).

The zeros of $f_a(\rho)$ can be calculated by using any standard routine, and zeros of $f_a(\omega)$ are obtained by using

$$\rho_i \xrightarrow{\mathcal{T}} \omega_i$$

here ρ_i s are the zeros of $f_a(\rho)$, and ρ is transformation of Equation (6.5).

6.3 Application and Discussion

Suppose we want to design a 2DFIR filter with transfer function defined as

$$|H(\omega)| = \begin{cases} 1000 & 0 \le \omega < 1.5908 \\ \text{transition band} & 1.5908 \le \omega < 2.0944 \\ 10 & 2.0944 \le \omega < \pi \end{cases}$$
(6.10)

Comparing these characteristics with Figure 6.2, we note that

$$A_{max} = 1000,$$
$$A_{min} = 10.$$

First we find the corresponding object function by using the transformation given in Equation (6.5). The object function comes out to be

$$f(\rho) = \begin{cases} 10 & 0 \le \rho < 0.5 \\ \text{transition band} & 0.5 \le \rho < 0.7 \\ 1000 & 0.7 \le \rho < 1.0 \end{cases}$$
(6.11)

Let us design this filter.

6.3.1 Design 1

To calculate the value of $f_a(\rho)$ (Equation (6.9)), we first calculate the values of a_i 's using Equation (6.8). As mentioned earlier, when we increase the number of terms to approximate our object function the approximation becomes better and better. Therefore, it is up to the designer where he wants to stop.

Let us approximate the object function using 8 orthogonal polynomials. We calculate the coefficients using Equation (6.8). The approximate object function is

$$f_a(\rho) = 750P_0 - 562.5P_2 - 468.75P_4 + 59.04P_6 + 263.6712P_8 + 306.152P_{10} + 66.65048P_{12} - 198.4407P_{14}$$
(6.12)

We apply Equations (6.3) and (6.6) on the approximated object function and get the filter in frequency domain. The magnitude response of the filter is shown in Figure 6.6, its cross section in Figure 6.7 and the magnitude response in dB is shown in Figure 6.8. When we look at Figures 6.6 and 6.7, we find that the pass band of the filter has some ripples. Figure 6.8 shows that first side band is approximately 35dB down.

In the next classification we consider the same problem while using more number of orthogonal polynomial terms to approximate the object function.

6.3.2 Design 2

In the present case we use 15 orthogonal polynomial terms to approximate the object function. This object function is then transformed using the transformation of Equation (6.5). The resulting 2DFIR is shown in Figure 6.9, the cross section in Figure 6.10 and the magnitude response in dB is shown in Figure 6.11, the first side band is approximately 45dB down.

Comparing Figures 6.7 and 6.10, it is clear that as we increase the number of terms to calculate object function, which in turn is used to construct the required filter, the sharpness of the transition band of the filter increases and ripples in the pass band decreases. The filter shown in Figure 6.11 has lower side bands than the filter represented in Figure 6.8.



Figure 6.6: Magnitude response of low pass FIR filter when object function is approximated using 8 orthogonal polynomial terms.

6.3.3 Design 3

We can easily design high pass, bandpass, band reject, and other type of filters using the same approach. We only have to calculate the coefficients required to design the filter.

Let us design a high pass filter with characteristics as below

$$|H(\omega)| = \begin{cases} 10 & 0 \le \omega < 2.0944 \\ \text{transition band} & 2.0944 \le \omega < 2.5322 \\ 1000 & 2.5322 \le \omega < \pi \end{cases}$$
(6.13)

The corresponding object function is

$$f(\rho) = \begin{cases} 1000 & 0 \le \rho < 0.3 \\ \text{transition band} & 0.3 \le \rho < 0.5 \\ 10 & 0.5 \le \rho < 1.0 \end{cases}$$
(6.14)

Suppose this object function is approximated using 15 orthogonal polynomial terms. The resulting high pass filter is shown in Figure 6.12, and



Figure 6.7: Cross section of the magnitude response of low pass FIR filter when object function is approximated using 8 orthogonal polynomial terms.



Figure 6.8: Magnitude response in dB of low pass FIR filter when object function is approximated using 8 orthogonal polynomial terms.



Figure 6.9: Magnitude response of low pass FIR filter when object function is approximated using 15 orthogonal polynomial terms.



Figure 6.10: Cross section of the magnitude response of low pass FIR filter when object function is approximated using 15 orthogonal polynomial terms.



Figure 6.11: Magnitude response in dB of low pass FIR filter when object function is approximated using 15 orthogonal polynomial terms.

the cross section of this filter is represented in Figure 6.13. From the figures it is clear that the filter has flat pass band and if we pass an image through it the filter will be able to cut off non required frequencies (we will pass some images through this filter in the next section). Phase remains linear and can easily be verified.

6.3.4 Design 4

Let us design a band pass filter with frequency domain characteristics

$$|H(\omega)| = \begin{cases} 1000 & 1.4455 \le \omega < 2.6362 \\ \text{transition bands} & 1.1096 \le \omega < 1.4455 \text{ and } 2.6362 \le \omega < 2.8405 \\ 10 & 2.8405 \le \omega < \pi \text{ and } 0 \le \omega < 1.1096 \end{cases}$$
(6.15)

The corresponding object function is



Figure 6.12: Magnitude response of high pass FIR filter when object function is approximated using 15 orthogonal polynomial terms.

$$f(\rho) = \begin{cases} 1000 & 0.25 \le \rho < 0.75 \\ \text{transition bands} & 0.15 \le \rho < 0.25 \text{ and } 0.75 \le \rho < 0.85 \\ 10 & 0.0 \le \rho < 0.15 \text{ and } 0.85 \le \rho < 1.0 \end{cases}$$
(6.16)

Using the procedure discussed and approximating the object function of Equation (6.16) with 15 polynomial terms we get the required filter characteristics, which are shown in Figure 6.14. The magnitude response in dB is shown in Figure 6.15, the pass band of filter is very flat and side bands are approximately 25 dB down.

6.4 Results

The results produced by the algorithm, proposed in this chapter, give promising results. By increasing the number of terms to approximate the object function we can design better filters; that is, filters with sharp cutoff, lower side band and less ripples in the pass band. This technique can



Figure 6.13: Cross section of magnitude response of high pass FIR filter when object function is approximated using 15 orthogonal polynomial terms.



Figure 6.14: Cross section of magnitude response of band pass FIR filter when object function is approximated using 15 orthogonal polynomial terms.



Figure 6.15: Full view of magnitude response in dB of band pass FIR filter when object function is approximated using 15 orthogonal polynomial terms.

easily be used to design low pass, high pass, band pass, band reject, and other type of filters.

If we pass some images through these filters, we can estimate the quality of our filter. Passing an image through a high pass filter means that the details of the image are kept, while the large scale gradients are removed, while in the case of low pass filter opposite of this happens.

Figure 6.16 shows an image which when passed through a high pass filter, designed using 15 orthogonal polynomial terms, becomes as shown in Figure 6.17, which depicts that only high frequency components (changes in the different gray levels of the image) are retained. When this image is passed through a low pass filter the resulting image is shown in Figure 6.18, which removes high frequency components (small details of image) of the image and only blurred details of the image are retained.

Similarly, an image with more details than the previous one is shown in Figure 6.19, its high pass filtered output is shown in Figure 6.20 and low pass in Figure 6.21, respectively.

A daily life image with a fair combination of changes is shown in Figure 6.22, its high pass filtered result is shown in Figure 6.23 and low pass filtered output is the Figure 6.24.

One more image is shown in Figure 6.25 its high pass and low pass filtered outputs are shown in Figures 6.26 and 6.27, respectively. We have shown various type of images to show that the filter designed can be used for any type of image. The filters used above are designed using 15 orthogonal polynomial terms to approximate the corresponding object function.



Figure 6.16: Image-1.



Figure 6.17: Image-1 passed through high pass filter designed using 15 orthogonal polynomial terms.



Figure 6.18: Image-1 passed through low pass filter designed using 15 orthogonal polynomial terms.



Figure 6.19: Image-2.

6.5 Conclusion

A simple new approach has been discussed here for designing 2D linear phase FIR filters. The cut off characteristics and ripples in the pass band can be controlled by using the method proposed in this chapter. As we increase the number of terms, the resulting filter becomes more flat in pass band and the transition region approximates the defined value. We can design any type of filter; that is, band pass, band reject, etc., using the present technique.



Figure 6.20: Image-2 passed through high pass filter designed using 15 orthogonal polynomial terms.


Figure 6.21: Image-2 passed through low pass filter designed using 15 orthogonal polynomial terms.



Figure 6.22: Image-3.



Figure 6.23: Image-3 passed through high pass filter designed using 15 orthogonal polynomial terms.



Figure 6.24: Image-3 passed through low pass filter designed using 15 orthogonal polynomial terms.



Figure 6.25: Image-4.



Figure 6.26: Image-4 passed through high pass filter designed using 15 orthogonal polynomial terms.



Figure 6.27: Image-4 passed through low pass filter designed using 15 orthogonal polynomial terms.

7 Design of 1 Dimensional Zero or Linear Phase IIR Filter with Orthogonal Polynomials

7.1 Introduction

In the past two to two and half decades a great deal of work has been carried out in the field of design of linear phase IIR filters. In general, design of exact linear phase IIR filter is not possible, schemes have been proposed to approximate pass band linearity. Conventionally, first the magnitude specifications of an IIR filter are met, and then all pass equalizers are applied to linearize the phase response [50,51]. Mostly IIR filters are designed with equiripple or maximally flat group delay [52]. But their magnitude characteristics are poor. Optimization techniques are used to simultaneously approximate magnitude and phase response characteristics [53,54]. To meet with the magnitude and phase characteristics at the same time, generally, linear programming is used [55]. Lu et al [56] gives an iterative procedure to directly design a linear phase IIR filter, it is based on a weighted least-squares algorithm. Xiao et al [57] discusses a method to design a linear phase IIR filter with frequency weighted least square error optimization using Broyden-Flether-Goldfarb-Shanno (BFGS) [58] method.

Model reduction approach has also been proposed by various authors [59,60]. A procedure to design linear phase IIR filter from linear phase FIR filter has been discussed by Holford *et al* [61] using frequency weighting model reduction, for highly elective filters. He [61] gives good compromise for order of the filter, pass band maximum ripple and stop band minimum attenuation. None of the techniques discussed above give perfect zero group delay IIR filter. The use of zero group delay filters makes it possible to obtain much higher data rates than with ordinary filters, since the rise time is extremely fast and there is less accumulated loss due to rise time [62,

63], various problems arising due to non zero group delay are disscussed in [64]. In the present chapter, we discuss a new technique to design the IIR filters having perfectly linear or zero phase response which consequently will lead to zero group delay. Group delay is defined as:

$$\tau_g = -\left|\frac{d\phi(\omega)}{d\omega}\right| \tag{7.1}$$

where, $\phi(\omega)$ is the phase as a function of frequency and ω is frequency in rad/s.

7.2 Preliminaries

Proposed zero group delay IIR filter design uses linear phase high pass and low pass FIR filters. To design a linear phase low pass IIR filter $(H_{iirlowpass}(\omega))$, we divide linear phase low pass FIR filter characteristics $(H_{LP(FIR)N}(\omega), \text{Figure 7.1})^{-1}$ with linear phase high pass FIR filter characteristics, $(H_{HP(FIR)D}(\omega), \text{Figure 7.2})^{-2}$. The required FIR filters can be designed by using the procedure discussed in Chapter 5. The magnitude characteristics of two FIR filters are such that the pass band of low pass FIR filter is the stop band of high pass FIR filter and vice-versa; the transition bands overlap, when one transition band goes from lower value to higher value other goes from higher to lower. Therefore, the low pass IIR filter function is given by

$$H_{iirlowpass}(\omega) = \frac{H_{LP(FIR)N}(\omega)}{H_{HP(FIR)D}(\omega)}$$

$$= \frac{|H_{LP(FIR)N}| \angle H_{LP(FIR)N}}{|H_{HP(FIR)D}| \angle H_{HP(FIR)D}}$$

$$= \frac{|H_{LP(FIR)N}|}{|H_{HP(FIR)D}|} \angle \left(H_{LP(FIR)N} - H_{HP(FIR)D}\right)$$
(7.2)

It is clear from Equation (7.2) that at every point the amplitudes of the low pass and high pass filters will be divided and phase subtracted, to get

¹The subscript indicates Low Pass FIR filter to be used as Numerator.

²The subscript indicates High Pass FIR filter to be used as Denominator.



Figure 7.1: Desired filter characteristics for LPF to be used as numerator.



Figure 7.2: Desired filter characteristics for HPF to be used as denominator.



Figure 7.3: Object function for LPF of Figure 7.1, showing slight non linearity in the transition region.

the amplitude and phase characteristics of the low pass IIR filter at these points, respectively. To calculate the pass band values of the IIR filter we divide pass band values of $|H_{LP(FIR)N}|$ with stop band values of $|H_{HP(FIR)D}|$ and vice versa to get the stop band. Because both, high pass and low pass, filters have linear phase, therefore, resulting IIR filter will have a zero phase (if the phase response is same for both of the numerator and denominator) or linear phase, Equation (7.2).

From Equation (7.2) we can deduce that while designing the FIR high pass filter; that is, $H_{HP(FIR)D}(\omega)$, we have to ensure that the filter has no zeros in $0 \le \omega \le \pi$ region, otherwise these zeros will make the resulting IIR filter unstable. If $H_{HP(FIR)D}(\omega)$ has zero crossings, then before designing the IIR filter we add a constant term (a term with zero frequency) to make sure that $H_{HP(FIR)D}(\omega)$ is above ω -axis; that is, there is no zero in $0 \le \omega \le \pi$. Also, because the phase of both of the FIR filters is linear the division will result in a linear phase IIR filter.

If our application requires a zero/constant phase IIR filter, then we can design such filter as well. It is clear from Equation (7.2) that to get zero phase we have to make sure that the phase characteristics of low pass and high pass FIR filters are identical; furthermore the linear phase

low pass FIR filter must not cross the ω -axis, because every such crossing results in a phase change of π . We can control magnitude as well as phase characteristics of resulting IIR filter, it will be discussed shortly. Since we need to have FIR filters to get the IIR filter; we have to design them first.

To design a low pass FIR filter ($H_{LP(FIR)N}(\omega)$), shown in Figure 7.1, we first have to construct a corresponding object function ($f_l(\omega)$), shown in Figure 7.3, as discussed and designed in Chapter 5. Similarly, we can design the high pass FIR filter, Figure 7.2, and its corresponding object function shown in Figure 7.4. Once we have the required FIR filters we move to design the IIR filter.



Figure 7.4: Object function for HPF of Figure 7.2, showing slight non linearity in the transition region.

Based on the method discussed above, we discuss the procedure to design the IIR filters in detail in the next section.

7.3 Procedure

Let us design a low pass, zero group delay IIR filter. To have a zero phase and consequently zero group delay, we have to make sure that pass/stop band of numerator and corresponding stop/pass band of denominator, and transition bands of both numerator and denominator must match. From here onwards we represent $H_{LP(FIR)N}(\omega)$ by $H_N(\omega)$ and $H_{HP(FIR)N}(\omega)$ by $H_D(\omega)$ for simplicity. To design such a filter we divide low pass FIR filter characteristics ($H_N(\omega)$), shown in Figure 7.1, by high pass FIR filter characteristics ($H_D(\omega)$), shown in Figure 7.2.

As discussed in the last section these FIR filter characteristics will be realized by their corresponding object functions – $f_N(x)$ for $H_N(\omega)$ and $f_D(x)$ for $H_D(\omega)$ – as shown in Figures 7.3 and 7.4, respectively. The object functions are given by

$$f_N(x) = \sum_{n=0}^{\infty} a_{2n} P_{2n}$$
(7.3)

$$f_D(x) = \sum_{n=0}^{\infty} b_{2n} P_{2n}$$
(7.4)

where, P_0 , P_2 , P_4 ... are the Legendre polynomials of the order of 0, 2, 4, ..., and a_{2n} and b_{2n} are coefficients which are multiplied with the Legendre polynomials to approximate the required object function characteristics shown in Figures 7.3 and 7.4, respectively. Only even terms of Legendre polynomials are used, considering the fact that filter characteristics in frequency domain are symmetrical in nature.

Following steps outline the procedure to design zero group delay IIR filters.

Step **1** : Because polynomials are orthogonal to each other we can calculate the coefficients $a_0, a_2, a_4, ...$ and $b_0, b_2, b_4, ...$ by using the following formulae³

$$a_{i} = \frac{\int_{0}^{1} f_{N}(x) P_{i} dx}{\int_{0}^{1} P_{i} P_{i} dx}$$
(7.5)

$$b_{i} = \frac{\int_{0}^{1} f_{D}(x) P_{i} dx}{\int_{0}^{1} P_{i} P_{i} dx}$$
(7.6)

where, i = 0, 2, 4, ...

Step 2 : We use finite number of a_i 's and b_i 's in Equations (7.3) and ³see Appendix A.

(7.4), therefore, we get the approximate object functions $f_{Na}(x)$ and $f_{Da}(x)$ in place of $f_N(x)$ and $f_D(x)$, respectively; that is,

$$f_{Na}(x) = \sum_{n=0}^{N} a_{2n} P_{2n}$$
(7.7)

$$f_{Da}(x) = \sum_{n=0}^{M} b_{2n} P_{2n}$$
(7.8)

Step 3 : The polynomials $f_{Na}(x)$ and $f_{Da}(x)$ are converted to $H_{Na}(\omega)$ and $H_{Da}(\omega)$, respectively, using the transformation discussed in Chapter 1 and repeated below for convenience

$$x = x_0 \cos(\omega/2) \tag{7.9}$$

We consider x_0 to be equal to 1, one can assume other values also.

The discussion in the previous section makes it clear that $H_{Na}(\omega)$ and $H_{Da}(\omega)$ represent the approximate LPF and HPF characteristics corresponding to $f_{Na}(x)$ and $f_{Da}(x)$, respectively.

Step 4 : Calculate the rational function in $cos(\omega/2)$

$$H_{iirlowpass}(\omega) = \frac{H_{Na}(\omega)}{H_{Da}(\omega)}$$
(7.10)

Notice that the denominator must not have a zero in $0 \le \omega \le \pi$ *and* $0 \le x \le \pi$ *, since this will lead to instability.*

Step 5 : We find the zeros of $H_{Na}(\omega)$ and $H_{Da}(\omega)$ by solving $H_{Na}(\omega) = 0$ and $H_{Da}(\omega) = 0$, respectively. They are represented by $z_{iN} = exp(j\omega_{iN})$ and $z_{iD} = exp(j\omega_{iD})$; where, i = 1, 2, 3, ..., and ω_{iN} and ω_{iD} 's are the zeros calculated.

Step 6 : Transfer function of the resulting IIR filter is

$$H_{iirlowpass}(z) = \frac{(z - z_{1N})(z - z_{2N})(z - z_{3N})\dots}{(z - z_{1D})(z - z_{2D})(z - z_{3D})\dots}$$
(7.11)

The transfer function given in the above equation is used to find out the magnitude response and phase response of our IIR filter.

To design high pass filter we have to divide high pass FIR filter characteristics by low pass FIR filter characteristics. For clear understanding of the above procedure we design some IIR filters using the procedure discussed above.

7.4 Application and Discussion

Suppose we intend to design an IIR filter with following characteristics

 A_{max} = 500 (pass band amplitude) A_{min} = 0 (stop band amplitude) ω_{OL} = 2.3186 (start of transition band) ω_{OH} = 2.0007 (end of transition band)

Therefore, as per the discussion of the previous section we have to design the low pass and high pass FIR filters first. The, assumed, filter characteristics are as follows

Low pass FIR (Figure 7.1) filter

$$A_{maxN} = 1000$$
$$A_{minN} = 0$$
$$\omega_{OL} = 2.3186$$
$$\omega_{OH} = 2.0007$$

High pass FIR (Figure 7.2) filter

$$A_{maxD} = 1$$

$$A_{minD} = 2$$

$$\omega_{OL} = 2.0007$$

$$\omega_{OH} = 2.3186$$

from the previous section it must be clear to the reader that above values will lead to zero phase IIR filter.

Note that as of now we show only positive half of the graph the negative half being a mirror image. Now we consider some specific cases in the next section.

7.4.1 Design 1

Suppose we take 10 Legendre polynomial terms to approximate the object functions (both for LPF and HPF characteristics). The object functions are approximated in the following manner

$$f_{Na}(x) = a_0 P_0 + a_2 P_2 + \ldots + a_{18} P_{18}$$
(7.12)

$$f_{Da}(x) = B + b_0 P_0 + b_2 P_2 + \ldots + b_{18} P_{18}$$
(7.13)

where, $f_{Na}(x)$ and $f_{Da}(x)$ are discussed in the previous section, *B* is added to the object function to make sure that the condition stated in Step 4 is met. According to Equations (7.3) and (7.4) we calculate the coefficients a_0, a_1, a_2, \ldots and b_0, b_1, b_2, \ldots . Then,

$$f_{Na}(x) = 530P_0 + 909.685P_2 - 586.385P_4$$

-4.64292P_6 + 401.108P_8 - 379.026P_{10} + ... (7.14)

$$f_{Da}(x) = 100 + 470P_0 - 909.685P_2 + 586.385P_4 +4.64292P_6 - 401.108P_8 + 379.028P_{10} + \dots$$
(7.15)

Here we have assumed the value of *B* as 100, one can assume any value of *B* provided the condition stated in Step 4 is satisfied.

Using the transformation (Equation (7.9)) we transform our object functions to approximate LPF and HPF characteristics; that is, $H_{Na}(\omega)$ and $H_{Da}(\omega)$, respectively. The zeros (in frequency domain) ω_{iN} and ω_{iD} 's corresponding to $H_{Na}(\omega)$ and $H_{Da}(\omega)$, as given in Step 5 and Step 6, are calculated next which in turn gives the transfer function $H_{iirlowpass}(z)$ as follows

$$H_{iirlowpass}(z) = \frac{\prod_{i=1}^{18} (z - z_{iN})}{\prod_{i=1}^{18} (z - z_{iD})}$$
(7.16)

where, z_i 's are $e^{j\omega_i}$; i = 1, 2...18.

This $H_{iirlowpass}(z)$ gives the magnitude response and magnitude response in dB, which are shown in Figures 7.5 and 7.6, respectively. The pass band of resulting filter has many ripples, although side bands are very low, but even then the design is not good. Let us approximate the object functions with more number of terms in the following section to have a better filter.



Figure 7.5: Magnitude response of low pass IIR filter designed using 10 orthogonal polynomial terms to approximate object function.

7.4.2 Design 2

Twenty Legendre polynomial terms are used to approximate the object functions (both for LPF and HPF characteristics) in the present example. After following the steps outlined in Section 2, we get the approximate object functions $f_{Na}(x)$ and $f_{Da}(x)$ and eventually $H_{iirlowpass}(\omega)$. The magnitude response, magnitude response in dB and phase response are shown in Figures 7.7, 7.8 and 7.9, respectively.

It becomes evident from the comparison of Figures 7.6 and 7.8 that the magnitude response becomes better in terms of flatness of pass band, and the second side band levels (they go almost 20dB further down when compared with the previous design). The transition band becomes sharper in the present design when compared with the previous design. In the same



Figure 7.6: Magnitude response in dB of low pass IIR filter corresponding to the object function approximated using 10 orthogonal polynomial terms.



Figure 7.7: Magnitude response of low pass IIR filter corresponding to the object function approximated using 20 orthogonal polynomial terms.



Figure 7.8: Magnitude response in dB of low pass IIR filter corresponding to the object function approximated using 20 orthogonal polynomial terms.



Figure 7.9: Phase response of low pass IIR filter corresponding to the object function approximated using 20 orthogonal polynomial terms.

fashion, if we increase the number of terms to approximate object functions further we get better and better results. The phase is zero and is shown in Figure 7.9.

7.4.3 Design 3

Hitherto we have considered that the order of object functions, corresponding to HPF and LPF, are same. Now we consider the case when the object function $f_{Na}(x)$ is synthesized using less number of Legendre polynomials than $f_{Da}(x)$. Suppose the order of $f_{Na}(x)$ is 10 and that of $f_{Da}(x)$ is 20, Figure 7.10 shows the magnitude response in dB.

Comparing Figures 7.8 and 7.10, we infer that if we decrease the number of terms to approximate numerator, side bands are not as lower as they were when both numerator and denominator were approximated using equal number terms. The transition band is also not that sharp in the present case, the reason becomes clear if we look at Equation (7.2). The phase, in this case, is linear.



Figure 7.10: Magnitude response in dB when numerator and denominator object functions are approximated using 10 and 20 orthogonal polynomial terms, respectively.

7.4.4 Design 4

In this design we consider that the object function $f_{Da}(x)$ is synthesized using 10 and $f_{Na}(x)$ using 20 orthogonal polynomial terms, respectively. The magnitude response in dB for this kind of IIR filter is shown in Figure 7.11.



Figure 7.11: Magnitude response in dB when numerator and denominator object functions are approximated using 20 and 10 orthogonal polynomial terms, respectively.

Figures 7.10 and 7.11 make it clear that as we increase the number of terms in the denominator, or in other words as we increase number of zeros in the denominator of our IIR filter, it has pass band which is more flat than of Design 4. The side band level is better if we design the IIR filter with less number of terms to approximate denominator object function (corresponding to high pass FIR filter). User has to decide, depending on application, which amongst them he would like to use. If we increase the number of terms used to approximate object function then definitely both type of filters will become better.

7.5 Conclusion

Above discussion makes it clear that an IIR filter with linear or zero phase and consequently zero group delay can easily be designed by using the orthogonal polynomials. The proposed IIR filter gives good cutoff characteristics. By increasing the number of polynomial terms, we can approximate our object function very closely, which in turn will produce good frequency characteristics both in the pass band and transition band. The ripples in the pass band becomes negligible as we increase the number of terms to approximate our object function. Second side band amplitudes decreases as we increase the number of terms in our object functions. In all, we may state that the alternate approach discussed in this chapter gives a much better design of IIR filter, with absolute zero group delay, when compared with the currently available methods [55, 57, 60, 65, 66], none of them have zero group delay.

B Design of 2 Dimensional Zero or Linear Phase IIR Filter with Orthogonal Polynomials

8.1 Introduction

Medical imaging [67], face recognition [68], and image processing [69] are some of the potential applications areas where 2 dimensional (2D) filters are used. Some applications; for example, multirate signal processing, and biomedical applications (ECG, EEG etc.) [70,71] need zero group delay characteristics coupled with good pass band characteristics. Huang *et al* [72] pointed out that for restoration of images, linear phase plays an important role. In the last few years various approaches have been discussed by various researchers to design a linear phase 2D IIR filter [73–77]. The procedures discussed in these papers either first design a 2D filter with desired amplitude response which is followed by 2D all pass transfer function to linearize the phase, or minimize the error of desired magnitude and phase response. Computer aided optimization techniques [78–80] are also used, but they are computationally very extensive.

Present method discusses a new approach which results in absolute linear phase and zero group delay 2D IIR filters without using any optimization.

8.2 Procedure

We extend the proposed idea of previous chapter to design 2D IIR filters with zero phase. To design such filters, first we need to calculate the 2D orthogonal polynomials (in cylindrical or ρ domain). These polynomials, as previously, will be used for realizing object functions. The procedure and the resulting polynomials are given in Chapter 6. To design the low pass IIR filter, we need to divide the low pass FIR filter characteristics with the high pass FIR filter characteristics, as discussed in the previous chapter and repeated below

$$H_{iirlowpass}(\omega) = \frac{H_{LP(FIR)\mathbf{N}}(\omega)}{H_{HP(FIR)\mathbf{D}}(\omega)}$$

= $\frac{|H_{LP(FIR)\mathbf{N}}| \angle H_{LP(FIR)\mathbf{N}}}{|H_{HP(FIR)\mathbf{D}}| \angle H_{HP(FIR)\mathbf{D}}}$
= $\frac{|H_{LP(FIR)\mathbf{N}}|}{|H_{HP(FIR)\mathbf{D}}|} \angle (H_{LP(FIR)\mathbf{N}} - H_{HP(FIR)\mathbf{D}})$ (8.1)

We proceed to design the object functions, $f_N(\rho)$ and $f_D(\rho)$, corresponding to numerator and denominator object functions respectively, as discussed in the previous chapter. ρ is in cylindrical co ordinate and is related with Cartisan co-ordinates (x,y) by $\rho^2 = x^2 + y^2$. For low pass and high pass filters (shown in Figures 8.1 and 8.2, respectively) example object functions are shown in Figures 8.3 and 8.4, respectively. Figures show half of the filter characteristics and object function, rest being mirror image of it.

From the previous chapter, we know that the object functions, $f_N(\rho)$ and $f_D(\rho)$, are weighted linear combinations of polynomials, and are represented as follows

$$f_{N}(\rho) = \sum_{m=0}^{\infty} a_{2m} P_{2m}(\rho)$$

$$f_{D}(\rho) = \sum_{m=0}^{\infty} b_{2m} P_{2m}(\rho)$$
(8.2)

where, P_{2m} are the orthogonal polynomials calculated in Chapter 6, and a_{2m} are the coefficients.

Because we use finite number of polynomials to design object functions, therefore, $f_N(\rho)$ and $f_D(\rho)$ become $f_{Na}(\rho)$ and $f_{Da}(\rho)$, respectively, and are given by



Figure 8.1: 3D desired low pass filter response.



Figure 8.2: 3D desired high pass filter response.



Figure 8.3: 3D object function for low pass filter of Figure 8.1, showing the slight non linearity in the transition region.



Figure 8.4: 3D object function for high pass filter of Figure 8.2, showing the slight non linearity in the transition region.

$$f_{Na}(\rho) = \sum_{m=0}^{N} a_{2m} P_{2m}(\rho)$$
(8.3)

$$f_{Da}(\rho) = \sum_{m=0}^{N} b_{2m} P_{2m}(\rho)$$
 (8.4)

The values of a_m and b_m are calculated by¹

$$a_{m} = \frac{\int_{0}^{1} f_{N}(\rho) P_{m}(\rho) d\rho}{\int_{0}^{1} P_{m}(\rho) P_{m}(\rho) d\rho}$$
(8.5)

$$b_{m} = \frac{\int_{0}^{1} f_{D}(\rho) P_{m}(\rho) d\rho}{\int_{0}^{1} P_{m}(\rho) P_{m}(\rho) d\rho}$$
(8.6)

The object functions – $f_{Na}(x, y)$ and $f_{Da}(x, y)$ or $f_{Na}(\rho)$ and $f_{Da}(\rho)$ – are transformed to filter characteristics – $H_{Na}(u, v)$ and $H_{Da}(u, v)$ or $H_N(\omega)$ and $H_D(\omega)$ – by using the transformation discussed previously and repeated below

$$\rho = \rho_0 \cos(\omega/2) \tag{8.7}$$

where, $\omega^2 = u^2 + v^2$ and we assume ρ_0 equal to 1.

The zeros of $f_{Na}(\rho)$ and $f_{Da}(\rho)$ can be calculated by using any standard routine, and zeros of $H_N(\omega)$ and $H_D(\omega)$ are obtained by using the transformation

$$\rho_j \xrightarrow{\mathcal{T}} \omega_j$$

where ρ_j s are the zeros of $f_a(\rho)$ and \mathcal{T} is transformation defined in Equation (8.7).

Next, we calculate the rational function by

$$H_{iirlowpass}(u,v) = \frac{H_{Na}(u,v)}{B + H_{Da}(u,v)}$$
(8.8)

B in Equation (8.8) is added to make sure that the division does not yield an infinite value and consequently an unstable filter, whenever $|H_{Da}(u, v)|$ crosses zero.

¹see Appendix A.

From the previous discussion we know that where $|H_{Da}(u, v)|$ has small values (in its stop band), $|H_{iirlowpass}(u, v)|$ will have large values (in its pass band). Very large oscillations in the pass band will correspond to a pass band which is not smooth. Thus to have a smooth pass band we need a sufficiently large value of *B*.

We can design a high pass filter by dividing high pass FIR filter characteristics by low pass FIR filter characteristics. If we are going to use same object functions then for high pass filter Equation (8.8) converts to

$$H_{iirhighpass}(u,v) = \frac{B + H_{Da}(u,v)}{H_{Na}(u,v)}$$
(8.9)

As discussed earlier to avoid zero crossing to occur in the denominator term, which is $|H_{Na}(u, v)|$, we have to add a constant term, say *A*. Thus, Equation (8.9) becomes

$$H_{iirhighpass}(u,v) = \frac{B + H_{Da}(u,v)}{A + H_{Na}(u,v)}$$
(8.10)

and we write Equation (8.8) as, assuming that same object functions are going to be used to realize both high pass and low pass filter,

$$H_{iirlowpass}(u,v) = \frac{A + H_{Na}(u,v)}{B + H_{Da}(u,v)}$$
(8.11)

From the previous discussion it is clear that *A* should be sufficiently large so as to make sure that there are no oscillations in the pass band. How different values of *A* and *B* will change the characteristics of resulting filter is shown in Table 8.1.

Type of Filter	А	В	Figure Number
Low Pass	0	1100	8.5
Low Pass	0	2000	8.6
Low Pass	2000	2000	8.7
High Pass	1100	0	8.8
High Pass	2000	0	8.9
High Pass	2000	2000	8.10

Table 8.1: Values of A and B vs. figures.

The stepwise procedure to design the low pass IIR filter is as follows *Step 1* : Calculate a_{2m} and b_{2m} .



Figure 8.5: Example low pass filter with A=0 and B=1100.



Figure 8.6: Example low pass filter with A=0 and B=2000.



Figure 8.7: Example low pass filter with A=2000 and B=2000.



Figure 8.8: Example high pass filter with A=1100 and B=0.



Figure 8.9: Example high pass filter with A=2000 and B=0.



Figure 8.10: Example high pass filter with A=2000 and B=2000.

Step 2 : Calculate the approximate object functions; that is, $f_{Na}(\rho)$ and $f_{Da}(\rho)$.

Step 3 : Calculate

$$H_{iirlowpass}(u, v) = \frac{A + H_{Na}(u, v)}{B + H_{Da}(u, v)}$$

Step 4 : Calculate the transfer function in *z* domain; that is, $H_{iirlowpass}(z)$.

Step 5 : Find the zeros of $H_{Na}(\omega)$ and $H_{Da}(\omega)$ represented by $z_{iN} = exp(j\omega_{iN})$ and $z_{iD} = exp(j\omega_{iD})$; where, i = 1, 2, 3, ..., and ω_{iN} and ω_{iD} 's are the calculated zeros.

Step 6 : The transfer function of the resulting low pass filter is given as

$$H_{iirlowpass}(z) = \frac{A + (z - z_{1N})(z - z_{2N})(z - z_{3N})\dots}{B + (z - z_{1D})(z - z_{2D})(z - z_{3D})\dots}$$
(8.12)

in the similar fastion we can design a high pass filter.

8.3 Application and Discussion

Let us design IIR filter with the following values:

Low pass FIR filter (to be used as numerator to design the IIR filters), Figure 8.11

$$A_{maxN} = 0dB$$
 (pass band magnitude)
 $A_{minN} = -60dB$ (stop band magnitude)
 $\omega_{OL} = 2.0944$
 $\omega_{OH} = 1.5908$

High pass FIR filter (to be used as denominator to design the IIR filters), Figure 8.12

> $A_{maxD} = 60 dB$ (pass band magnitude) $A_{minD} = 0 dB$ (stop band magnitude) $\omega_{OL} = 1.5908$ $\omega_{OH} = 2.0944$



A = 2000,

Figure 8.11: Desired filter characteristics for LPF.

8.3.1 Design 1

Let us design an IIR filter with 9 Legendre polynomial terms used to approximate both the object functions; that is, to calculate $f_{Na}(\rho)$ and $f_{Da}(\rho)$. Following the procedure discussed in the previous section we design the resulting filter.

Figures 8.13 and 8.14 represent the designed IIR low pass and high pass filters, respectively. Figures 8.15 and 8.16 show the low and high pass filter response in dB. It is clear from the Figures 8.13 and 8.15 that the pass band of the resulting IIR filter has oscillations in the pass band and stop band. Similarly, if we look at Figures 8.14 and 8.16 we find the same.

When an image (Figure 8.17) is passed through the low pass IIR filter, we have designed using 9 Legendre polynomial terms, the output is shown

and



Figure 8.12: Desired filter characteristics for HPF.



Figure 8.13: Magnitude response of low pass IIR filter when object functions are approximated using 9 orthogonal polynomial terms.



Figure 8.14: Magnitude response of high pass IIR filter when object functions are approximated using 9 orthogonal polynomial terms.



Figure 8.15: Magnitude response in dB of low pass IIR filter when object functions are approximated using 9 orthogonal polynomial terms.


Figure 8.16: Magnitude response in dB of high pass IIR filter when object functions are approximated using 9 orthogonal polynomial terms.

in Figure 8.18 and if it is passed through the high pass filter then the output is represented in Figure 8.19. In Figure 8.18 we can clearly see that the high frequency components; that is, the changes or edges of the original image, are lost and only the slow changes or low frequency components are intact. Figure 8.19 shows that only the high frequency components (small, sharp details) of the image are present and low frequency components are lost.

8.3.2 Design 2

Let us calculate approximate object functions using 15 polynomial terms and then design the filter. The resulting IIR low pass filter is shown in Figure 8.20. Similarly, if we design a high pass filter using 15 polynomial terms then the filter is shown in Figure 8.21.

From Figures 8.20 and 8.21 it becomes evident that there is an improvement in the pass band and stop band characteristics of the filter. If we draw the frequency characteristics in dB, Figures 8.22 and 8.23, then the improvement can be seen clearly. The pass band is very flat and has less



Figure 8.17: Image-1.

oscillations, and also the stop band is lower than in the previous case.

Passing the previous image, Figure 8.17, through this filter we get Figures 8.24 and 8.25 resulting from high pass and low pass filtering, respectively. The images make it clear that increment in the number of terms to approximate the object function improves the filter performance. The difference between the images of Figures 8.18 and 8.25 is very small and in general we can not make out the difference in one look. We have shown one such difference by encircling it in Figures 8.18 and 8.25. The human eye behaves as a low pass filter, therefore it is very difficult to see any difference between images of Figures 8.19 and 8.24, the high passed results, through naked eyes.

If we filter an image with greater details (like an image having text) through our 15th order high pass filter, the resultant images can show the quality of our filter in a better way. Figure 8.26 is such an image while Figure 8.27 shows the high pass filtered output. It is evident from the high pass filtered image, Figure 8.27, that it is clearer than the original image; that is, we can view the text better. Figures 8.28 and 8.29 shows the closer



Figure 8.18: Image-1 passed through low pass filter designed using 9 orthogonal polynomial terms.



Figure 8.19: Image-1 passed through high pass filter designed using 9 orthogonal polynomial terms.



Figure 8.20: Magnitude response of low pass IIR filter when object functions are approximated using 15 orthogonal polynomial terms.



Figure 8.21: Magnitude response of high pass IIR filter when object functions are approximated using 15 orthogonal polynomial terms.



Figure 8.22: Magnitude response in dB of low pass IIR filter when object functions are approximated using 15 orthogonal polynomial terms.



Figure 8.23: Magnitude response in dB of high pass IIR filter when object functions are approximated using 15 orthogonal polynomial terms.



Figure 8.24: Image-1 passed through high pass filter designed using 15 orthogonal polynomial terms.

Figure 8.25: Image-1 passed through low pass filter designed using 15 orthogonal polynomial terms.

view of a part of these images shown in Figures 8.26 and 8.27, respectively.

भागणे द्यायनम् ॥ ७०१ म ॥ ध्युध्वं। ईत्रे। पुरः ६६तं। प्रतस्य। हेवं। कुलि ॥होतरि स्तुअ्पातंमं॥ अभिनः (इदेक्तिः) कुर्षि अम्। १३में । कदनिः छन्। मा। हे ना ध्या। १३६ । पुस्ति। ध्युध्वन्ते। हर्षि : अन्यतं वर्षि अम्। १३में । कदनिः छन्। मा। हे ना ध्या। १३६ । पुस्ति। ध्युध्वन्ते। हर्षि : अन्यतं ॥ परि, क्य अम्हि । इत्य परिकृत्मे ॥ अन्यि। पुस्ति अ खुरा दिश्वर्याः परि, क्य आस्ति। सः। इत्य हर्षि छतिः ज्युध्वि अत्ति। कुवि अस्ति । स्याः चित्रक्र्ययः स्तमः। हवः । हर्षे जान् द्य १ ॥ यत्र भ्या स्तुर्धाः स्त्रि अर्थाः भ्यास्थ्या भ्यास्थ मत्रे। इत्य स्त्राः प्र रा १ ॥ यत्र भ्या स्तुर्धाः स्त्रा दिव अर्द्धाः हार्षा द्या स्था भ्यास्थ भयाः प्र रा १ ॥ यत्र भ्या स्तुर्धाः स्त्रा दिव अर्द्धाः हार्षा द्या स्यान्य भ्यास्थ भयां भ्याः प्र रा ३ भयत्र भ्याः सत्रिक्ताः हार्षा द्यास्य भ्यास्थ भयाः भ्यान्य भयाः भ्याः भ्याः भ्याः स्तात्र अत्य स्त्राणीः भाषां। अन्यस्थ हार्या द्या भ्यान्य भयाः भ्यात्र स्त्राः भ्याः भ्याः स्त्रत्वे। ज्यात्रा स्तुर्धा युन् । स्तु सर्वस्वानः श्यरत्ये ॥ २ भाषा स्ति । सः। स्त्राः स्त्राः स्त्राः स्त्रा

द्वीतभिष्यम्भभौगाः। अर्द्धस्ताः ॥त्रषोभ्याहिः खुभि भहवे॥ सर्थासर्ति । उन्हेभ्यस्य त्रोध्यप्रके अस्तिर्गः अक्षुसः सामास्य भावयभ्य देदायुर्हिति स्वयम् अर्थ्यत्राः जिलाति। सः खुर्भे ॥ इत्त्रवासिर्ग्व भावयभ्य देवायुर्हिते स्वयम् अत्राभा भये आभ्यत्ते ॥ देवनः भो ॥ उद्याति भिरू भायास्ति । इंदे ॥ आस्ति स्वयः अत्राभा । वाक्षि सद्धिवार्ग्तती प्रदेशः भो ॥ उद्याति भिरू भायास्ति । इंदे ॥ आस्ति स्वयः अत्राभी । वाक्षि सद्धिवार्ग्तती प्रयम् ॥ तो भ्याभ्यात् अभ्य स्वयः भ्रा समिद्धार्त्तः इंदे भन्न ॥ ज्याः भाति । उर्थ ॥ तो भ्याभ्यात् अभ्याः वियागः भाषा वियाय् इत्य भ्याः स्वत्य अत्राभ्यात् । त्राभ्यात् भ्याः अत्यां ॥ स्वयः भ्रा भवित्य भ्याः भ्यान् भ्याः भ्यान्य स्वयः भ्याः भ्यात् भ्याः भ्यात् भ्याः भ्यात्ताः भ्यात्त्वः भ्याः भ्यायः भ्याः भ्यायाः भ्याः भ

Figure 8.26: Image-2.

If we apply our high pass filter on an ultrasound image of a fetus, Figure 8.30, the result is shown in Figure 8.31. Another image which is of human bone structure and its high pass filtered output is shown in Figures 8.32 and 8.33, respectively. The filtered output show that the edges present in the original image are enhanced and therefore it is easier to find different parts of fetus in Figure 8.30 and defects, if there is any, from the bone structure.

Next we pass a satellite image through our IIR filter. The satellite images have a lot of high frequency details (like outlines of the street of a city, river, buildings, etc.). Figure 8.34 shows the original satellite image and when it is passed through 2DIIR filter the outcome is shown in Figure 8.35. The edges of the original image are retained in the resulting image.

म्त्रीगणे शासनमः॥ ७०६ म.॥ जुर्कते। मुँछे। पुरः अर्थि। पुरास्तं। दुवं। कुसि अहातरि कितुआ के पि आजुर्कि। पूर्वि भि अद्रिपि मि अद्रिपि कि अद्रसि अत्रसि आतु मि या जा। दुर्ग पुस्तानि॥ जुर्कते पुर्वि अद्रमि अद्रपि मि अद्रपि अत्रसि आतु कि या जा। दुर्ग पुस्तानि॥ जुर्कते पुर्वि अपू व्युक्त प्रति क्रुप्ति अपू वि अपू ति वा रक्ता अद्रुग पुस्तानि। जुर्क अत्र तुरं। जुन्दर्य ते अपू वि अपू अर्थि अत्र कि वा रक्ता अत्र प्रति अत्र तुरं अपू खुरं। जिन्द्रम् या जिन्द्रम् अत्र कि अपू अर्थ वा रक्ता अत्र प्रति अत्र कि अत्र तुरं। सामा जिन्द्रम् या जिन्द्रम् अत्र कि अपू अर्थ आता अत्र प्रति अत्र कि अत्र तुरं अप्र तुरं अत्र कि अपूर्व अत्र कि अपू कि अत्र कि अपू अर्थ अत्र भाषा के अत्र प्रति कि अत्र कि जो दे राष्ट्र दिया कि कि वि अत्र कि अपू अप्र इंगा उप त्या अपू अत्र कि अत्र कि जो दे राष्ट्र वि आत्र कि अत्र कि अत्र कि अत्र कि अत्र कि अत्र कि अत्र कि उन्हे की अत्र कि जुन्द्र अर्थ की का कि कि अत्र इंगा उप त्या अत्र कि जिन्द्र अर्थ कि कि कि अत्र कि अत्र कि आ क्या कि अप्र कि अर्थ कि अत्र अत्र कि अत्र अत्र कि अत्र अत्र कि अत्र कि कि अत्र अत्र कि कि अत्र कि कि अत्र कि अत्र कि अत्र कि कि अत्र कि अत्र कि अत्र कि अत्र कि अत्र कि कि अत्र कि कि अत्र कि अत्र कि अत्र कि कि अत्र कि क्र कि अत्र कि कि अत्र कि

द्र्श्तिभिः द्रमेश्मीमां अस्त्रे कलाशाल्मीत्युक्ति न्युपि एच गर्भावत्यि उ का भिः द्र्गिक्ति अस्तिता क्रुक्त समिश- क्रुक्त अपि एच गर्भात् कि उ का भिः द्रान्तानि सन्दर्भ भुक्त समिश- क्रुक्त अपि क्रिया क्रुक्ति में द्र्या क्रुक्ट्रिया क् उनगानि सन्दर्भ भुक्त क्रिक्त कि प्रात्म अर्थ देवायू हर्गि ग्रान् क्रुक्ट्रिया प् जार्भ्य में गर्द्दन क्रिक्त क्रुक्ति क्रिक्ता क्रुक्ति द्रिया क्रुक्ति प् स्तुत्विया क्रिक्त क्रिक्त क्रिक्ता क्रुक्ति क्रिक्त भुक्त क्रिक्त क्रिक्त क्रिक्त स्तुत्विया क्रिक्त क्रिक्त क्रिक्त क्रिक्त क्रिक्त क्रिक्त क्रिक्त क्रुक्ति क्रुक्ति क्रिक्त अन्ता भावता क्रिक्त क्रिक्त क्रिक्त क्रिक्त क्रुक्त क्रिक्त क्रिक्त क्रिक्त क्रिक्त क्रिक्त कर्मत क्रान्त्य क्रिक्त कर्मत क्रान्त्य क्रिक्त क्रि

Figure 8.27: Image-2 passed through high pass filter designed using 15 orthogonal polynomial terms.

TI men C

Figure 8.28: Image-3.

8.4 Conclusion

Above discussion makes it clear that an IIR filter with zero phase and consequently zero group delay can easily be designed by using the orthogonal polynomials. The IIR filter designed above gives good cutoff characteristics. By increasing the number of polynomial terms, we can approximate our object function very closely, which in turn will produce good frequency characteristics both in the pass band and the transition region. The ripples in the pass band becomes negligible as we increase the number of terms to approximate our object function. Stop band amplitude decreases as we increase the number of terms in our object functions. One can use these filters for specific purposes like, satellite images, medical image enhancement also. In all, we may state that the alternate approach discussed in this chapter gives a much better design of IIR filter, with absolute zero group delay, when compared with the currently available methods.

Figure 8.29: Image-3 passed through high pass filter designed using 15 orthogonal polynomial terms.

Figure 8.30: Image-4.

Figure 8.31: Image-4 passed through high pass filter designed using 15 orthogonal polynomial terms.

Figure 8.32: Image-5.

Figure 8.33: Image-5 passed through high pass filter designed using 15 orthogonal polynomial terms.

Figure 8.34: Image-6

Figure 8.35: Image-6 passed through high pass filter designed using 15 orthogonal polynomial terms.

9 Conclusion and Future Work

In this dissertation, we have proposed design methodologies to design the linear phase, 1D and 2D, FIR and IIR digital filters. Due to the simplicity of the procedure and excellent magnitude and phase characteristics, filters designed in the present work have a broad range of applications [2]. Most standard methods deal only either on the magnitude or phase characteristics, whereas, we have proposed algorithms which produce digital filters with user defined magnitude characteristics with the linear phase. The methods discussed give us unique design in terms of magnitude characteristics. Method to design zero group delay IIR filters is also discussed.

The proposed 1D and 2D Chebyshev FIR filter design methodology can be used to realize filters having equripple side bands, and for designing narrow to very narrow band filters with side band level defined by the user. Narrow band filters are very useful if we want to remove a particular frequency (it may be due to noise or in case of image representing a particular color or gray level) from the signal spectrum. The bandwidth of the filter can be controlled by using the modified approach to design the Chebyshev filters. When a satellite image is passed through this type of high pass filter edges (high frequency components of an image) were well retained.

We have proposed 1D and 2D FIR and IIR filter design based on orthogonal polynomials. It can be used for designing digital filters with linear phase and used specific magnitude characteristics. This type of design technique can be used to design multiband, notch, low pass, high pass, band pass and band reject digital filters. Depending on the application requirements we can design required type of filters.

We have passed different types of images through proposed 2D filters and shown the simulated results. Medical images, satellite images, text as images, everyday life images, and other type of images are passed through these digital filters. The filtered images show the quality of our filters. When an ultrasound image is passed through our 2DIIR digital filter, it is enhanced by fairly good amount. If one realizes application specific digital filter, the results will be promising. Satellite images were passed and a better and enhanced outcome was received. The filter design parameters are less in number, therefore, computational requirements are less.

There are various fields where present approach can be extended. Adaptive filters have various applications [81], one can extend the concept presented in the present thesis to design such filters. Development of video processing filters is another area where present technique can be extended. One can design a wavelet filter bank, designing wavelet using polynomials, which can be used to generate very narrow band to wide band filters. The discussed 2D filters can be used in conjunction with edge detecting algorithm to enhance edges of an image.

A Appendix

The sum of Legendre polynomials multiplied by suitable coefficients approximates the ideal polynomial in the least mean square error sense.

Proof :

Suppose a polynomial f(x) is approximated by $f_a(x)$ and its representation is

$$f_a(x) = \sum_{n=0}^{N} a_{2n} P_{2n}(x)$$

The mean squared error (MSE) polynomial between the original polynomial and approximated polynomial is given by

$$E(x) = \{f(x) - f_a(x)\}^2$$

or,

$$E(x) = \left[f(x) - \sum_{n=0}^{N} a_{2n} P_{2n}(x) \right]^2$$
(A.1)

Note that E(x) is either positive or zero hence the minimum value of its integral is zero. We define

$$\epsilon(a_0, a_2, \ldots) = \int_{-1}^{1} E(x) dx = \int_{-1}^{1} \left[f(x) - \sum_{n=0}^{N} a_{2n} P_{2n}(x) \right]^2 dx$$
 (A.2)

We need to find the coefficients $a_0, a_2, ...$ from the above equation so that this integral is minimized. The general way of solving this problem is well known, and is described below

$$\frac{\partial}{\partial a_i} \int_{-1}^{1} \left[f(x) - \sum_{n=0}^{N} a_{2n} P_{2n}(x) \right]^2 dx = 0 \quad i = 1 \dots N$$
(A.3)

or

$$\frac{\partial}{\partial a_i} \int_{-1}^{1} \left[\{f(x)\}^2 + \left\{ \sum_{n=0}^{N} a_{2n} P_{2n}(x) \right\}^2 -2f(x) \left\{ \sum_{n=0}^{N} a_{2n} P_{2n}(x) \right\} \right] dx = 0i = 1 \dots N$$
(A.4)

which is

$$\int_{-1}^{1} \left[0 + \frac{\partial}{\partial a_i} \left\{ \sum_{n=0}^{N} a_{2n} P_{2n}(x) \right\}^2 -2 \frac{\partial}{\partial a_i} f(x) \left\{ \sum_{n=0}^{N} a_{2n} P_{2n}(x) \right\} \right] dx = 0i = 1 \dots N$$
(A.5)

spelt out this is

$$\int_{-1}^{1} \left[\frac{\partial}{\partial a_i} \left\{ a_0^2 P_0^2 + 2a_0 a_2 P_0 P_2 + \dots + a_i^2 P_i^2 + 2a_i a_{i+2} P_i P_{i+2} + 2a_i a_{i+4} P_i P_{i+4} + \dots \right\} dx$$
$$-2 \int_{-1}^{1} f(x) a_i P_i \right] dx = 0$$
(A.6)

The orthogonality property of the Legendre polynomials reduces this integral to

$$2\int_{-1}^{1} a_i P_i^2 dx - 2\int_{-1}^{1} f(x) P_i dx = 0 \quad i = 1...N$$
 (A.7)

$$a_{i} = \frac{\int_{-1}^{1} f(x) P_{i} dx}{\int_{-1}^{1} P_{i}^{2} dx} \qquad i = 1 \dots N$$
(A.8)

B References

- [1] J. Alan V.Oppenheim, Ronald W.Schafer, *Discrete-Time Signal Processing*. Prentice Hall of India Pvt. Ltd., 2001.
- [2] A.Antoniou, *Digital Filters Analysis, Design, and Applications*. Tata McGraw Hill, 1999.
- [3] R. G. Lyons, *Understanding Digital Signal Processing*. Pearson Education India, fourth reprint ed., 2002.
- [4] A. Fettweis, "Digital circuits and systems," *Circuits and Systems, IEEE Transactions on*, vol. 31, pp. 31–48, Jan 1984.
- [5] T.W.Parks and J.H.McClellan, "Chebyshev-approximation for nonrecursive digital filters with linear phase," *IEEE Transactions on Circuit Theory*, vol. CT-19, pp. 189–194, 1972.
- [6] T. J.H.McClellan and L.R.Rabiner, "A computer program for designing optimum fir linear phase digital filters," *IEEE Transactions on Audio Electroacoustics*, vol. AU-21, pp. 506–526, 1973.
- [7] R. F. J.K.Liang and F.C.Lu, "Design of optimal nyquist, partial response, nth band, and nonuniform top spacing fir digital filters using linear programming techniques," *IEEE Transactions on Circuits Systems*, vol. CAS-32, no. 4, pp. 386–392, 1985.
- [8] A. W. Potchinkov, "Design of optimal linear phase fir filters by a semiinfinite programming technique," *Signal Processing*, vol. 58, no. 2, pp. 165–180, 1997.
- K.Steiglitz, "Computer aided design of recursive digital filters," *IEEE Transactions on Audio and Electroacoustics*, vol. AU-18, pp. 123–129, 1970.

- [10] J.W.Bandler and B.J.Bardakjian, "Least pth optimization of recursive digital filters," *IEEE Transactions on Audio and Electroacoustics*, vol. AU-21, pp. 460–470, 1973.
- [11] K. Preuss, "A novel approach for complex chebyshev-approximation with fir-filters using the remez exchange algorithm," in *Proceedings of ICASSP-87*, vol. 2, pp. 872–875, 1987.
- [12] K. Preuss, "On the design of fir filters by complex chebyshev approximation," *IEEE Transactions Acoustics, Speech and Signal Processing*, vol. 37, pp. 702–712, 1989.
- [13] T.W.Parks and J.H.McClellan, "A program for the design of linear phase finite impulse response digital filters," *IEEE Transactions on Audio and Electroacoustics*, vol. AU-20, pp. 195–199, 1972.
- [14] D. J. Shpak and A. Antoniou, "A generalized remkz method for the design of fir digital filters," *IEEE TRANSACTIONS ON CIRCUITS AND SYSTEMS*, vol. 37, no. 2, pp. 161–174, 1990.
- [15] J. L.R.Rabiner and T.W.Parks, "Fir digital filter design techniques using weighted chebyshev approximation," *Proceeding of the IEEE*, vol. 63, pp. 595–610, 1975.
- [16] A.Antoniou, "New improved method for the design of weighted chebyshev, nonrecursive, digital filters," *IEEE transactions on Circuits and Systems*, vol. CAS-30, pp. 740–750, 1983.
- [17] E.C.Jordan and K.G.Balman, *Electromagnetic Waves and Radiating Systems*. Prentice Hall of India Pvt. Ltd., 2 ed., 2003.
- [18] T. W. Parks and J. H. McClellan, "Chebyshev approximation for nonrecursive digital filters with linear phase," *IEEE Transactions on Circuit Theory*, vol. CT-19, pp. 189–194, 1972.
- [19] T. P. J. McClellan, "A united approach to the design of optimum fir linear-phase digital filters," *IEEE Transactions on Circuits Theory*, vol. 20, no. 6, pp. 697–701, 1973.
- [20] L.R.Rabiner and B.Gold, *Theory and Application of Digital Signal Processing*. Prentice Hall of India Pvt. Ltd., 1999.

- [21] A. E.M.Hofstetter and J. Siegel, "A new technique for the design of non-recursive digital filters," *In Proc. Fifth Annual Princeton Conference on Information Sciences and Systems*, pp. 64–72, 1971.
- [22] T. I.Fotinopoulos, "Equiripple minimum phase fir filter design from linear phase systems using a novel technique for polynomial factorisation," in *IEEE Global Telecommunications Conference*, 2000. GLOBECOM '00., vol. 3, pp. 1659 – 1663, 2000.
- [23] V. J.M.A.Tanskanen, "Round-off error free fixed-point design of polynomial fir predictors," in *Conference Record of the Thirty-Third Asilomar Conference on Signals, Systems, and Computers, 1999.*, vol. 2, pp. 1317 – 1321, 1999.
- [24] J. G. Proakis and D. G. Manolakis, *Digital Signal Processing: Principles, Algorithms, and Applications*. Prentice Hall International, third ed., 1996.
- [25] A.G.Constantinides, "Spectral transformations for digital filters," Proceedings of the IEEE, vol. 117, pp. 1585–1590, 1970.
- [26] J. Moorer, "General spectral transformations for digital filters," IEEE Transactions on Acoustics, Speech, and Signal Processing, vol. 29, pp. 1092–1094, 1981.
- [27] K. Swamy, M.N.S.; Thyagarajan, "Frequency transformations for digital filters," *Proceedings of the IEEE*, vol. 65, no. 1, pp. 165–166, 1977.
- [28] D.E.Dudgeon and R. Mersereau, Multidimensional Digital Signal Processing. Prentice Hall, 1984.
- [29] T.W.Parks and C.S.Burrus, *Digital Filter Design*. Wiley International, 1987.
- [30] L. G. Cuthbert, "Optimizing nonrecursive digital filters to nonlinear phase characteristics," *The Radio and Electronic Engineer*, vol. 44, pp. 645–651, 1974.
- [31] J.H.McCallig, "Design of digital fir filters with complex conjugate pulse responses," *IEEE Trabsactions on Acoustics, Speech and Signal Processing*, vol. CT-20, pp. 697–701, 1973.

- [32] E.Y.Remes, "General computation methods of tchebycheff approximation," *Tech. Rep.*, pp. 1–85, 1957.
- [33] J. A. A. Holt and R. Bennett, "Iterative technique for designing nonrecursive digital filters to nonlinear phase characteristics," *The Radio* and Electronic Engineer, vol. 46, pp. 589–592, 1976.
- [34] L. M. D. I. Barrodale and J. C. Mason, "Linear chebyshev approximation of complex-valued functions," *Mathematics of Computation*, vol. 32, pp. 853–863, 1978.
- [35] J. V. Hu and L. R. Rabiner, "Design techniques for two-dimensional digital filters," *IEEE Transactions on Audio Electroacoustics*, vol. AU-20, pp. 249–257, 1972.
- [36] J. G. Fiasconaro, "Two-dimensional nonrecursive digital filters," *Picture Processing and Digital Filters, T. S. Huang*, vol. Springer-Verlag, 1975.
- [37] W.-S. Lu, "A unified approach for the design of 2-d digital filters via semidefinite programming," *IEEE Transactions on Circuits System*, vol. 49, pp. 814–826, 2002.
- [38] W. Lu and T. Hinamoto, "A new minimax design for 2-d fir filters with low group delay," in ISCAS2005, Kobe, Japan, p. 2028âĂŞ2031, May, 2005.
- [39] J. A. B. C. S. Burrus and I. W. Selesnick, "Iterative reweighted leastsquares design of fir filters," *IEEE Transactions Signal Processing*, vol. 42, no. 11, pp. 2926–2936, 1994.
- [40] S. C. Chan and K. M. Tsui, "On the design of real and complex fir filters with flatness and peak error constraints using semidefinite programming," in *Proceedings of IEEE International Symposium. on Circuits* and Systems., pp. 125–128, 2004.
- [41] M. T. Hanna, "Design of linear phase fir filters with a maximally flat passband," IEEE Transactions on Circuits and Systems-11: Analog and Digital Signal Processing, vol. 43, no. 2, pp. 142–147, 1996.

- [42] E.Kreyszig, Introductory Functional Analysis with Applications. John Wiley and Sons, 8 ed., 2005.
- [43] J.Lim, Two-Dimensional Signal and Image Processing. Prentice Hall, 1990.
- [44] C. Charalambous, "The performance of an algorithm for minimax design of two-dimensional linear phase fir digital filters," *IEEE Transaction on Circuits and Systems*, vol. 32, pp. 1016–1028, 1985.
- [45] W.-S. Lu and A. Antoniou, *Two-Dimensional Digital Filters*. Marcel Dekker, 1992.
- [46] J.H.McClellan, "The design of two dimensional digital filters by transformations," in *Proc. 7th Annual Princeton Conference on Information Sciences*, pp. 247–251, 1973.
- [47] T. J.H.McClellan and L.R.Rabiner, "A computer program for designing optimal fir linear phase digital filters," *IEEE Transactions on Audio and Electroacoustics*, vol. AU-21, No.6, pp. 506–526, 1973.
- [48] W. R.M.Mersereau and T. Jr., "Mcclellan transformation for twodimensional digital filtering: I-design & ii-implementation," *IEEE Transactions on Circuit and Systems*, vol. CAS-23, no. 4, pp. 414–422, 1976.
- [49] S. Haykin, *Communication System*. John Wiley and Sons, 3rd ed., 1996.
- [50] A. Antoniou, *Digital Filters: Analysis, Design, and Applications*. McGraw-Hill, "2nd" ed., 1993.
- [51] C. Charalambous and A. Antoniou, "Equalization of recursive digital filters," *IEE Proceedings Part G*, vol. 127, no.4, pp. 219–225, 1980.
- [52] J.P.Thiran, "Recursive digital filters with maximally flat group delay," *IEEE Transactions on Circuit Theory*, vol. CT-18, pp. 659–664, 1971.
- [53] G. Cortelazzo and M. R. Lightner, "Simultaneous design in both magnitude and group delay of iir and fir filters based on multiple criterion optimization," *IEEE Transactions on Acoustics, Speech and Signal Processing*, vol. ASSP-32, pp. 949–967, 1984.

- [54] H. Baher, "Digital filters with finite-band approximation to constant amplitude and delay, and arbitrary selectivity," *Proceeding of the IEEE International Symposium on Circuits ans Systems*, pp. 657–659, 1986.
- [55] A. T. Chottera and G. A. Jullien, "A linear programming approach to recursive digital filter design with linear phase," *IEEE Transactions on Circuits and Systems*, vol. CAS-29, pp. 139–149, 1982.
- [56] W.-S. L. S.-C. P. C.-C. Tseng, "A weighted least-squares method for the design of stable 1-d and 2-d iir digital filters," *IEEE Transactions on Signal Processing*, vol. 46, pp. 1–10, 1998.
- [57] J. C. O. Chengshan Xiao and P. Agathoklis, "Design of linear phase iir filters via weighted least-squares approximation," *Proceedings of ICASSP*, vol. 6, pp. 3817–3820, 2001.
- [58] R. Fletcher, *Practical Methods of Optimization*. JohnWiley & Sons, 2nd ed., 1987.
- [59] I. K. B. Beliczynski and G. D. Cain, "Approximation of fir by iir digital filters: an algorithm based on balanced model reduction," *IEEE Transactions on Signal Processing*, vol. 40, pp. 532–542, 1992.
- [60] G. A.-R. M. F. Fahmy, Y. M. Yassin and N. El-Gayed, "Design of linear phase iir filters from fir specifications," *IEEE Transactions on Signal Processing*, vol. 42, pp. 437–440, 1994.
- [61] S. Holford and P. Agathoklis, "The use of model reduction techniques for designing iir filters with linear phase in the passband," *IEEE Transactions on Signal Processing*, vol. 44, pp. 2396–2404, 1996.
- [62] H. Taub and D.L.Schilling, *Principles of Communication System*. Tata McGraw Hill, 2nd ed., 2002.
- [63] H.R.Walker, "Systems involving group delay." www.vmsk.org/NRZGroupDel.pdf, March 2004.
- [64] P. J. Pupalaikis, "Group delay and its imserial data transmission testing." and pact on www.lecroy.com/tm/library/registerPDF.asp?wp=946, 2006.

- [65] V. Sreeram and P. Agathoklis, "Design of linear-phase iir filters via impulse-response gramians," *IEEE Transactions on Signal Processing*, vol. 40, pp. 389–394, 1992.
- [66] W.-Y. Y. Luowen Li, Lihua Xie and Y. C. Soh, "Design of low-order linear-phase iir filters via orthogonal projection," *IEEE Transactions on Signal Processing*, vol. 47, pp. 448–457, 1999.
- [67] R. Kreuger, "Virtex-em fir filter for video applications." Available: http://direct.xilinx.com/bvdocs/appnotes/xapp241.pdf.
- [68] S. M. S. Gong and A. Psarrou, "Dynamic vision: From images to face recognition," *1st ed. Imperial College Press*, 2000.
- [69] R. Gonzalex and R. Woods, *Digital Image Processing*. Prentice Hall, 2001.
- [70] A.F.Kohn, "Phase distortion in biological signal analysis caused by linear phase fir filters," *Medical and Biology Engineering and Computing*, vol. 25, pp. 231–238, 1987.
- [71] P.K.Sadasivan and D. Dutt, "Use of finite wordlength fir digital filter structures with improved magnitude and phase characteristics for reduction of muscle noise in eeg signals," *Medical and Biology Engineering and Computing*, vol. 33, pp. 306–312, 1995.
- [72] J. B. T. S. Huang and A. G. Deczky, "The importance of phase in image processing filters," *IEEE Transaction on Acoustics, Speech and Signal Processing*, vol. ASSP-23, pp. 529–542, 1975.
- [73] B. A. S. G. Gu and C. Zhang, "Synthesis of 2-d linear phase digital filters," *IEEE Transaction on Circuits and Systems*, vol. 37, pp. 1499–1508, 1990.
- [74] G. Gu and B. Shenoi, "A novel approach to the synthesis of recursive digital filters with linear phase," *IEEE Transactions on Circuits and Systems*, vol. 38, pp. 602–612, 1991.
- [75] H. K. Kwan and C. L. Chan, "Multidimensional spherically symmetric recursive digital filter design satisfying prescribed magnitude and

constant group delay response," *IEE Proceeding*, vol. part G, pp. 187–193, 1987.

- [76] K. Nishikawa and R. M. Mersereau, "Design of circularly symmetric two-dimensional iir lowpass digital filters with constant group delay using mcclellan transformation," *IEICE Transaction*, vol. E75-A, pp. 830–836, 1992.
- [77] R. Unbehauen, "Recursive digital lowpass filters with maximally flat group delay and chebyshev stopband attenuation," *Proceeding IEEE International Symposium on CIrcuits and Systems*, pp. 593–596, 1980.
- [78] S. A. H. Aly and M. M. Fahmy, "Design of two-dimensional recursive digital filters with specified magnitude and group delay characteristics," *IEEE Transactions on Circuits and Systems*, vol. CAS-25, pp. 908–916, 1978.
- [79] A. T. Chottera and G. A. Jullien, "Design of two-dimensional recursive digital filters using linear programming," *IEEE Transactions on Circuits and Systems*, vol. CAS-29, pp. 817–826, 1982.
- [80] G. A. Maria and M. M. Fahmy, "A new design technique for recursive digital filters," *IEEE Transactions on Circuits and Systems*, vol. CAS-23, pp. 323–325, 1976.
- [81] M. Bellanger, *Adaptive Digital Filters*. Marcel Dekker Ink, New York, 2001.

C Vita

EDUCATION	
Doctor of Philosophy,	Pursuing
In the field of Digital Signal and Image Processing	
Electronics and Communication Engineering Department	
Solan, H.P.	
Guide: Prof. Sunil Bhooshan.	
Master of Technology,	2002
Faculty of Engineering,	
Dayalbagh Educational Institute, Dayalbagh	
Guide: Prof V.P.Pyara.	
Bachelor of Engineering,	2000
Electronics & Communication Engineering,	
Dr. K.N.Modi Institute of Engineering & Technology, Modinagat	r.

SHORT TERM COURSES

- 1. Certificate Course on Embedded Systems & Applications Indian Institute of Technology, New Delhi.
- French Language Course Alliance Francaise de Delhi, New Delhi.

PROJECTS

Post Graduation Projects

- Recognition of Continuous Hindi Speech Signals using HMMs (Major Project)
 Place: D.E.I., Faculty of Engineering , Dayalbagh, Agra.
- Study of vector quantization and its various applications (Minor Project)
 Place: D.E.I., Faculty of Engineering , Dayalbagh, Agra.

3. Image denoising using wavelet transform(Minor Project) Place: D.E.I., Faculty of Engineering , Dayalbagh, Agra.

Under graduation Projects

- 1. Natural language processing using semantic analyzer (Major project) Place: Dr.K.N.Modi Institute of Engineering & Tech., Modinagar.
- Salary structure (Minor project)
 Place: Dr.K.N.Modi Institute of Engineering & Tech., Modinagar.

PAPERS PUBLISHED

- 1. Vinay Kumar and Sunil Bhooshan, "A new class of zero group delay recursive digital filters", submitted to IEEE Transactions on Signal Processing, 2008.
- 2. Sunil Bhooshan, Vinay Kumar, "Design of Two Dimensional Linear Phase Chebyshev FIR Filters", 9th, IEEE IET International Conference on Signal Processing, October 26-29, 2008 (accepted).
- Sunil Bhooshan, Vinay Kumar, A.Raghavendra Kumar, Rohini Mukhopadhyay, P.Varun Choudary, Meenu Jain and S.Bharat Kumar, "Design of Linear Phase 2 Dimensional FIR Filter: A New Approach", IEEE IET International Conference on Audio, Language and Image Processing, July 7-9, 2008, pp- 1359-1364.
- Sunil Bhooshan, Vinay Kumar, S. Mukerjee, "Modeling the Transient Response of a Hysteretic Medium: A Signal Processing Approach", IEEE Congress on Image and Signal Processing, May 28-30, 2008, pp-107-110.
- Sunil Bhooshan, Vinay Kumar, Ujjwal Verma, Hitesh Vatsyayan, Kumar Rohit, "T-Law: A New Suggestion for Signal Companding", IEEE Congress on Image and Signal Processing, May 28-30, 2008, pp-190-194.
- Sunil Bhooshan, Vinay Kumar, "A New Method to DesignÂăLinear Phase FIR Filter", IEEE International Conference on Industrial Technology, April 21-24, 2008, pp- 1-6.

- Sunil Bhooshan and Vinay Kumar, "Design of Chebyshev FIR Filter Based On Antenna Theory Approach", Proceedings of the 6th WSEAS International Conference on Signal Processing, Robotics and Automation, Corfu Island, Greece, February 16-19, 2007, pp-97-101.
- Sunil Bhooshan and Vinay Kumar, "A Novel Approach Towards The Design Of Chebyshev FIR Filter With Linear Phase", WSEAS Transactions on Signal Processing, Issue 2, Volume 3, February 2007, pp- 179-185.
- Sunil Bhooshan and Vinay Kumar, "A Polynomial Approach Towards the Design of Linear Phase FIR Filters", IEEE International Conference on Industrial Technology, December 15-17, 2006, pp- 632-636.
- Shalini Priti, Saurabh Shyam Mittal, Udai Singh, and Vinay Kumar, "Approximation by Matrix Mean of Walsh Fourier Series", Advances in Mathematics Research, Volume 5, 2005, pp- 31-45.
- 11. Vinay Kumar, "Hidden Markov modeling for Speech Recognition", Operation Research Society of India, 2004.

WORK EXPERIENCE

- Jaypee University of Information and Technology (2005-till now) Working as a senior lecturer in the Department of Electronics and Communication Engineering , teaching "Digital Signal and Speech Processing" and "Signal Processing Techniques", and "Digital Image Processing".
- Indreprastha Engineering College (2003-2005)
 Worked as a lecturer in the Department of Electronics and Communication Engineering, taught "Digital Communication", and "Digital Signal Processing".